AFWL-TR-71-57

2 20 7347

A TURBULENT MODEL STUDY FOR THE MULTICOMPONENT NONSIMILAR TURBULENT BOUNDARY LAYER PROGRAM

Larry W. Anderson Howard L. Morse

Aerotherm Corporation

TECHNICAL REPORT NO. AFWL-TR-71-57

October 1971

AIR FORCE WEAPONS LABORATORY Air Force Systems Command Kirtland Air Force Base New Mexico

NOV 8 1971 C

Reproduced by NATIONAL TECHNICAL INFORMATION SERVICE Springfield, Va. 22151

Approved for public release; distribution unlimited.

UNCLASSIFIED			
DOCUMENT CONT	ROL DATA - R	s D	
(Security classification of title, body of abetract and indexing .	ennotation must be a	ntered when the	overall report is classified)
Aerotherm Corporation		UNCLAS	SIFIED
Mountain View, California 94040			
A TURBULENT MODEL STUDY FOR THE MULTICOMPOLAYER PROGRAM	ONENT NONSIM	ILAR TURBU	LENT BOUNDARY
A DESCRIPTIVE NOTES (Type of report and inclusive dates) May 1970 through March 1971		989 in - University of Landson	
Larry W. Anderson Howard L. Morse			
6. REPORT DATE	74. TOTAL NO. OF	PAGES	75, NO. OF REFS
October 1971	206		94
E29601-70-C-0055	98. ORIGINATOR'S	REPORT NUMB	ER(S)
b. PROJECT NO. 5791	AFWL-TR-7	71-57	
^{c.} Task 27	95. OTHER REPOR this report)	T NO(5) (Any off	her numbers that may be assigned
<i>d.</i>			
Approved for public release; distribution	unlimited.	ILITARY ACTIV	11TY
	AFWL (SYI Kirtland	AFB, NM 8	37117
13. ABSTRACT (Distribution Limit	ation Staten	nent A)	
An investigation of the accuracy of the Air lent Boundary Layer Computer Code through co data is presented. Five complete data sets originally considered over one hundred separ investigations. The five data sets include with and without acceleration, supersonic fl distributions, subsonic flow with various wa wall blowingBefore the present study, the tively untested for compressible flows, but comparisons with other popular turbulent mod five data sets are carried out, and approxim which include velocity, total temperature, a stations, plus momentum thickness and drag co tion. Overall agreement between the profile good for flow with and without blowing. Dra typically below reported values for the data taken as inconclusive since there is disagree research community as to the validity of the the computer code for turbulent boundary lay option, is also presented. The entropy laye entropy gradients caused by shock curvature conditions.	Force Weapor omparisons of are selected rate turbuler flat plate s ow with high all blowing r e turbulent m is left esse lels. Detail mately 75 gra und Mach numb coefficient v e predictions g coefficient sets select estent among ese data. In rer predictions r model dire with the bou	is Laborato prediction from a line to boundary supersonic ly nonsimi- rates, and wodel in the entially un- ed predict phical com- per profile ariations and the en- the turbul formation rs, and or ectly coupl- indary laye	bry Nonsimilar Turbu- ons with experimental iterature survey which a layer experimental flow, hypersonic flow that wall temperature supersonic flow with the AFWL code was rela- there after detailed tions for each of the mparisons are presented as at several axial along the flow direc- experimental data is for for blown flows is fer, these results are lent boundary layer on the correct use of a new entropy layer tes the inviscid flow er edge boundary
DD FORM 1473		UNCLAS	SIFIED

Security Classification



AIR FORCE WEAPONS LABORATORY Air Force Systems Command Kirtland Air Force Base New Mexico 87117

When US Government drawings, specifications, or other data are used for any purpose other than a definitely related Government procurement operation, the Government thereby incur no responsibility nor any obligation whatsoever, and the fact that the Government may have formulated, furnished, or in any way supplied the said drawings, specifications, or other data, is not to be regarded by implication or otherwise, as in any manner licensing the holder or any other person or corporation, or conveying any rights or permission to manufacture, use, or sell any patented invention that may in any way be related thereto.

This report is made available for study with the understanding that proprietary interests in and relating thereto will not be impaired. In case of apparent conflict or any other questions between the Government's rights and those of others, notify the Judge Advocate, Air Force Systems Command, Andrews Air Force Base, Washington, DC 20331.

DO NOT RETURN THIS COPY. RETAIN OR DESTROY.

Security Classification			
REV +0403	LINE A	LINE O	LINE C
Boundary layer Turbulent Turbulent model			
			entre anno 1991 - Lance - Ir ear a-Arte - Arte
	Normality of the second s		An - Harry M. Kar, - a - M a - Andreas
	vr. try t		

1

Ì

Security Classification

i

A TURBULENT MODEL STUDY FOR THE MULTICOMPONENT NONSIMILAR TURBULENT BOUNDARY LAYER PROGRAM

Larry W. Anderson Howard L. Morse

Aerotherm Corporation

TECHNICAL REPORT NO. AFWL-TR-71-57

Approved for public release; distribution unlimited.

ABSTRACT

(Distribution Limitation Statement Λ)

An investigation of the accuracy of the Air Force Weapons Laboratory Nonsimilar Turbulent Boundary Layer Computer Code through comparisons of predictions with experimental data is presented. Five complete data sets are selected from a literature survey which originally considered over one hundred separate turbulent boundary layer experimental investigations. The five data sets include flat plate supersonic flow, hypersonic flow with and without acceleration, supersonic flow with highly nonsimilar wall temperature distributions, subsonic flow with various wall blowing rates, and superscnic flow with wall blowing. Before the present study, the turbulent model in the AFWL code was relatively untested for compressible flows, but is left essentially unchanged after detailed comparisons with other popular turbulent models. Detailed predictions for each of the five data sets are carried out, and approximately 75 graphical comparisons are presented which include velocity, total temperature, and Mach number profiles at several axial stations, plus momentum thickness and drag coefficient variations along the flow direction. Overall agreement between the profile predictions and the experimental data is good for flow with and without blowing. Drag coefficient prediction for blown flows is typically below reported values for the data sets selected. However, these results are taken as inconclusive since there is disagreement among the turbulent boundary layer research community as to the validity of these data. Information on the correct use of the computer code for turbulent boundary layer predictions, and on a new entropy layer option, is also presented. The entropy layer model directly couples the inviscid flow entropy gradients caused by shock curvature with the boundary layer edge boundary conditions.

iii/iv

FOREWORD

This report was prepared by the Aerotherm Corporation, Mountain View, California, under Contract F29601-70-C-0055. The research was performed under Program Element 62301F, Project 5791, Task 27.

Inclusive dates of research were May 1970 through March 1971. The report was submitted 1 September 1971 by the Air Force Weapons Laboratory Project Officer, Sergeant Dennis W. Lankford (SYT).

This technical report has been reviewed and is approved.

mis (1) Lanklor DENNIS W. LANKFORD

Sergeant USAF Project Officer

JÄY R. ROLAND Major USAF Chief, Theoretical Branch

Filkon

CARL B. HILLAND Lt Colonel USAF Chief, Technology Division

CONTENTS

Section		Page
I	INTRODUCTION	1
II	 COLLECTION AND EVALUATION OF TURBULENT BOUNDARY LAYER DATA 1. Summary of Experimental Data Considered 2. Criteria Used for Data Selection 3. Selected Data Sets a. Coles - Supersonic Flow With Zero Pressure Gradient b. NOL - Hypersonic Flow With and Without Pressure Gradient c. TRW - Hypersonic Flow With Step in Wall Temperature d. Stanford - Zero Pressure Gradient Subjonic Flow With and Without Uniform Blowing, and With a Step in Blowing e. Jeromin - Supersonic Flow With and Without Uniform Blowing 	3 3 13 14 15 19 19 20 21
III	<pre>TURBULENT MODEL SPLECTION 1. Description of Leading Turbulent Models a. Aerotherm Model b. Cebeci - Smith Model c. Bushnell - Beckwith Model 2. Turbulent Model Comparisons for the Law of the Wall Region a. WALAW Program b. Results for Three Near Wall Models 3. The Selected Turbulent Model a. Wall Law b. Wake Law c. Turbulent Prandtl and Schmidt Number</pre>	23 24 30 36 39 40 63 65 65
IV	PREDICTIONS AND COMPARISONS WITH DATA 1: Presentation of Comparisons a. Coles Comparisons b. NOL Comparisons c. TRW Comparisons d. Stanford Comparisons e. Jeromin Comparisons J. Rationalization of Results a. Overview of Agreement Between Experiment and Theory b. Changes in the Turbulent Model c. Desirability of Turbulent Model Changes for Blown Flows	68 68 69 106 120 135 150 152 152
V	CONCLUSIONS AND RECOMMENDATIONS	163
Appendix	- General Discussion of Computer Code Setup and Options	167
Referenc	285	183

ILLUSTRATIONS

Figure		Page
l	Velocity Ratio Profiles - Wieghardt, Subsonic Flow, Zero Pressure Gradient	26
2	Linear-Log Velocity Ratio Profile - Wieghardt, Subsonic Flow, Zero Pressure Gradient	27
3	Momentum Thickness Reynolds Number vs. Streamwise Location - Wieghardt, Subsonic Flow, Zero Pressure Gradient	28
4	Skin Friction Coefficient vs. Streamwise Location - Wieghardt, Subsonic Flow, Zero Pressure Gradient	29
5	Velocity Ratio Profiles - Kendall, Subscnic Flow, $F = 0.00325$	31
6	Linear-Log Velocity Ratio Profile - Kendall, Subsonic Flow, F = 0.00325	32
7	Momentum Thickness Reynolds Number vs. Streamwise Location Kendall, Subsonic Flow, F = 0.00325	33
8	Skin Friction Coefficient vs. Streamwise Location - Kendall, Subsonic Flow, $F = 0.00325$	34
9	Variation of Damping Constant for Bushnell-Beckwith Wall Model	37
10	Variation of Mixing Length for Bushnell-Beckwith Wake Model	38
11	Mixing Length Variation with No Blowing	41
12	Mixing Length Variation with $F = 0.004$	42
J 3	Mixing Length Variation with $F = 0.010$	43
14	Linear-Log Velocity Ratio Profiles - Wieghardt, Subsonic Flow, Zero Pressure Gradient	44
15	Linear-Log Velocity Ratio Profile - Coles Run #20, Supersonic, Zero Pressure Gradient Flow	45
16	Linear-Log Mach Number Profile - NOL, Lee, et al., Run 12902, Hypersonic, Zero Pressure Gradient Flow	46
17	Linear-Log Static Temperature Profile - NOL, Lee, et al., Run 12902, Hypersonic, Zero Pressure Gradient Flow	47
18	Linear-Log Mach Number Profile - NOL, Lee et al., Run 2611, Hypersonic, Zero Pressure Gradient Flow	48
19	Linear-Log Static Temperature Profile - NOL, Lee, et al., Run 2611; Egpersonic Coro Pressure Gradient Flow	49
20	Linear-Log Velocity Ratio Profile - MIT, Kendall, Run C - 2 x 10^{-3} - 50, Low Speed Flow with Blowing	51
21	Linear-Log Velocity Ratio Profile - MIT, Kendall Run C - 3 \times 10 ⁻³ - 50, Low Speed Flow with Blowing	52
22	Linear-Log Velocity Ratio Profile - MIT, Kendall Run C - 5 x 10^{-3} -50, Low Speed Flow with Blowing	53
23	Linear-Log Velocity Ratio Profile - Stanford, Simpson Run 2367-3. Low Speed Flow without Blowing	55

California California

.

ILLUSTRATIONS (continued)

Figure		Page
24	Linear-Log Velocity Ratic Profile - Stanford, Simpson Run 122366-2, Low Speed Flow with Blowing	56
25	Linear-Log Velocity Ratio Profile - Stanford, Simpson Run 122066-3, Low Speed Flow with Blowing	57
26	Linear-Log Velocity Ratio Profile - Stanford, Simpson Run ' 121966-4, Low Speed Flow with Blowing	58
27	Linear-Log Velocity Ratio Profile - Jeromin Run 2.5-0.0-1.40, Supersonic Flow without Blowing	59
28	Linear-Log Velocity Ratio Profile - Jeromin Run 2.5-1.2-1.40, Supersonic Flow with Blowing	60
29	Linear Log Velocity Ratio Profile - Jeromin Run 3.5-0.0-1.40, Supersonic Flow without Blowing	61
30	Linear-Log Velocity Ratio Profile - Jeromin Run 3.5-2.1-1.40, Supersonic Flow with Blowing	62
31	Effect of Constant in Wake Region - Eddy Viscosity Expression	66
32	Effect of Turbulent Prandtl Number on Recovery Factor at M = 3.7	67
33	Mach Number Ratio Profile - Coles Run #27, Supersonic, Zero Pressure Gradient Flow	70
34	Velocity Ratio Profile - Coles Run #27, Supersonic, Zero Pressure Gradient Flow	71
35	Linear-Log Velocity Ratio Profile - Coles Run #27, Supersonic, Zero Pressure Gradient Flow	72
36	Mach Number Ratio Profile - Coles Run #20, Supersonic, Zero Pressure Gradient Flow	73
37	Velocity Ratio Profile - Coles Run #20, Supersonic, Zero Pressure Gradient Flow	74
38	Linear-Log Velocity Ratio Profile - Coles Run #20, Supersonic, Zero Pressure Gradient Flow	75
39	Momentum Thickness Reynolds Number vs Streamwise Location - Coles' Runs at Mach 2.6 and 3.7	76
40	Skin Friction Coefficient vs Streamwise Location - Coles' Run at Mach 2.6 and 3.7	77
41	Mach Number Ratio Profiles - NOL, Lee, et al., 5 Atm, Hypersonic, Zero Pressure Gradiant Flow	79
42	Total Temperature Ratio Profiles - NOL, Lee, et al., 5 Atm, Hypersonic, Zero Pressure Gradient Flow	80
43	Velocity Ratio Profiles - NOL, Lee, et al., 5 Atm, Hypersonic, Zero Pressure Gradient Flow	81
44	Linear-Log Velocity Ratio Profiles - NOL, Lee, et al., 5 Atm., Hypersonic, Zero Pressure Gradient Flow	82

vii

3502-13-11-57

ILLOSTRATIONS (constinued)

Figure		Page
45	Momentum Ministers Republic Sumber We Streamwice Location - NOL. Let. et al., 5 Ann, Repersonic, More Pressure Studient Flow	23
24	Skin Friction Coefficient ve Streamise Location - 302, Lee, et al., 5 Max Sypersonic, Lero Freesure Gradient Flow	1 4
47	Mach Sumber Ratio Frofiles - SOL, Gee, et al., 10 Atm, Sypersonic. Agro Fressure Gradient Flow	\$ 5
42	Total Temperature Ratio Profiles - MRN, Los, et al., 19 Ana, Hypersonic, Tero Pressure La Allent Flow	Ħ
4 3	Aclocity Batic Frofiles - 982, Lee, et al., 13 Ann, Experionit, Lero Fressite Gradient Flow	27
92	Minear-Mog Velocity Batic Profiles - SOL, Lee, et al., 18 Abs, Nypersonic, Jero Pressure Madient Flow	Ŧ3
51	Armenium Thickness Reynolds Number of Streamonse Location - SCE. Les. et al., 10 Alm. Eggensonic. Ser) Pressure Gradient Fick	82
72	Saim Frieddon Opefficient we Streamwise Locating - SDL. Lee. Et n., 10 Ann. Rypersonic. Sono Fressure Gradient Flow	50
33	Mach Number Ratio Profiles - NOC, Brotz, et al., 5 Atm. Sypersonic, Megaline Processe Gradient Flow	34
34	Total Temperature Batic Froniles - SOL, Spott, et al., 5 Ata, Sypersonic, Megatuwe Freesure Fradient Flow	7 5
55	Velocity Batio Profiles - MML, Brott, et al., 5 Ann. Appersonic, Negative Pressury Cradient Flow	74
54	lineur-log Velocity Latic Profiles - WOL, Stort, et al., 5 Atm, Hypersonic, Wegative Pressure Readient Flow	47
50	Romatum Shickness Republics Number vs Streamwise Location - MOL, Bruty et al., - Aim, Hypersonur, Megazive Fressure Gradient Flow	35
55	Skim Friction Coefficient vs Streamwise Location - MOL, Brott, et al., 3 Ann. Experisonic, Separive Pressure Scadient Flow	53
57	Mach Summer Actic Frofiles - NOL, Brott, et al., 10 Atm, Hypersonic, Meganive Pressure Gradient Flow	140
5 0	Total Numperature Ratus Profiles - NOL, Srott, et al., 20 Atm, Rypersonus, Negative Preasure Grazient Flow	161
ŧī	Velocity Ratio Frofiles - MLL, Brott, et al., 10 Aum, Rypersonic, Negative Pressure Stadient Flow	162
62	Linear-Log Velocity Ratio Profiles - MCL, Brott, et al., 10 Atm, Rypersonic, Segative Pressure Gradient NLOW	123
63	Momentum Thickness Reymolds Number as Streamsise Location - NGL, Brott, et al., 11 Atm, Hypersonic, Sepative Pressure Gradient Flow	204
54	Skin Frietich Coefficient vs Striamuse Lovation - MOL, Brobt, et al., 15 Ann, Syperstnic, Negative Pressure Gradient Flow	190

Î

ATML-TR-71-57

ì

1.1.1.1.1.1.1

an a star for the set of the set

:

of poster

- - -

•

ILLUSTRATIONS (continued)

Figure		Page
65	Wall Temperature vs X, Distance from Leading Edge of Cylinder TRN Step-Op in Wall Temperature	108
66	Nach Bumber Rztio Profiles - TRN, Step-Up in Wall Temperature, Supersonic	109
67	Total Temperature Ratic Profiles - TRW, Step-Up in Wall Temperature, Supersonic	110
68	Velocity Catio Profiles - TRN, Step-Up in Wall imperature, Supersonic	111
69	Linear-Log Velocity Ratio Profiles - TRW, Step-Up in Wall Temperature, Supersonic	112
70	SFin Friction Coefficient vs Streamwise Location - TRW, Step-Up in Wall Temperature, Supersonic	113
71	Wall Temperature vs X, Distance from Leading Edge of Cylinder, TRN, Step-Down in Wall Temperature	114
72	Mach Number Ratio Profiles - TRW, Step-Down in Wall Temperature, Supersonic	115
73	Total Temperature Ratio Profiles - TRW, Step-Down in Wall Temperature, Supersonic	116
74	Velocity Ratio Profiles - TRW, Step-Down in Wall Temperature, Supersonic	117
75	Linear-Log Velocity Ratic Profiles - TRW, Step-Down in Wall Temperature, Supersonic	118
76	Skin Friction Coefficient vs Streamwise Location - TRW, Step-Down in Wall Temperature, Supersonic	119
רץ	Velocity Ratio Profiles - Stanford, Simpson Run #2367, Subsonic, No Blowing	121
78	Linear-Log Velocity Ratio Profiles - Stanford, Simpson Run #2367, Subsonic, No Blowing	122
7 9	Velocity Ratic Profiles - Stanford, Simpson Run #122366, Subscric, F = 0.002	123
80	Linear-Log Velocity Ratio Profiles - Stanford, Simpson Run #122366, Subsonic, P = 0.002	124
81	Velocity Ratic Profiles - Stanford, Simpson Run #122066, Subsonic, F = 0.004	125
82	Linear-Log Velocity Ratio Profiles - Stanford, Simpson Run #122056, Subsonic, F = 0.004	126
83	Velocity Ratio Profiles - Stanford, Simpson Run #121966, Subsonic, P = 0.008	127
84	Linear-Log Velocity Ratio Profiles - Stanford, Simpson Run #121966, Subsonic, F = 0.008	128

Į

ix

and an and the second

Contraction of the

ILLUSTRATIONS (continued)

÷

÷

Figure		Page
83	Velocity Ratio Profiles - Stanford, Simpson Run #5867, Subsonic, Step-Up in Blowing	129
86	Linear-Log Velocity Ratio Profiles - Stanford, Simpson Run #5867, Step-Up in Blowing	130
87	Momentum Thickness Reynolds Number vs Streamwise Location - Stanford, Uniform Blowing Runs, Subsonic	131
88	Momentum Thickness Reynolds Number vs Streamwise Location - Stanford, Simpson Run #5867, Subsonic, Step-Up in Blowing	132
89	Skin Friction Coefficient vs Streamwise Location - Stanford, Uniform Blowing Runs, Subsonic	133
90	Skin Friction Coefficient vs Streamwise Location - Stanford, Simpson Run #5867, Subsonac, Step-Up in Blowing	134
91	Velocity Ratio Profiles - Jeromin Run 2.5-0.0, Supersonic, No Blowing	137
92	Linear-Log Velocity Ratio Profiles - Jeromin Run 2.5-0.0, Supersonic, No Blowing	138
93	Velocity Ratio Profiles - Jeromin Run 2.5-1.2, Supersonic, F = 0.0012	139
94	Linear-Log Velocity Ratio Profiles - Jeromin Run 2.5-1.2, Supersonic, F = 0.0012	140
95	Momentum Thickness Reynolds Number vs Streamwise Location - Jeromin Mach 2,5 Runs	141
96	Skin Friction Coefficient vs Streamwise Location - Jeromin Mach 2.5 Runs	142
97	Velocity Ratio Profiles - Jeromin Run 3.6-0.0, Supersonic, No Blowing	143
98	Linear-Log Velocity Ratio Profiles - Jeromin Run 3.6-0.0, Supersonic, No Blowing	144
99	Velocity Ratio Profiles - Jeromin Run 3.6-2.1, Supersonic, F = 0.0021	145
100	Linear-Log Velocity Ratio Profiles - Jeromin Run 3.6-2.1, Supersonic, F = 0.0021	146
101	Momentum Thickness Reynolds Number vs Streamwise Location - Jeromin Mach 3.6 Runs	147
102	Skin Friction Coefficients vs Streamwise Location - Jeromin Mach 3.6 Runs	148
103	Static Temperature Profiles, Jeromin, M = 3.6, F = 0.000	151
104	Diagram Showing Law of the Wall Nomenclature	153
105	Comparison of Predicted Skin Friction Coefficient for Zero Blow- ing Stanford Case to Data and Empirical Values	154

F

*

ILLUSTRATIONS (concluded)

Figure		Page
106	Effect of Wall Law Changes on Drag Coefficient in Unblown Flat Plate Flow	156
107	Effect of Wall Law Changes on Velocity Profile in Unblown Flat Plate Flow	157
108	Effect of Wall Law Changes on Velocity Profile in Flat Plate Flow With Blowing	158
109	Effect of Wall Law Changes on Drag Coefficient in Flat Plate Flow With Blowing	159
110	Comparison of Various Theories With Simpson Data, Taken From Reference 89	160
111	Comparison of Blowing Correction From Several Investigators, Taken From Reference 87	162
112	η - ξ Coordinate System	168
113	Comparison of BLIMP Velocity Profile Predictions for Different Numbers of Nodes	170
114	Typical Velocity and Velocity Gradient Profiles in a Turbulent Boundary Layer	171
115	Overshoot of the Spline Fit Procedure	172
116	Definition of Near Wall Velocity Profile for Data Start Input	176
117	Response of Velocity Profile Following a Data Start; Example, Jeromin 2.5-1.2	177
118	a _H Constraint With Entropy Layer Flow	182

TABLES

Table		Page
I	Experimental Data With No Blowing	4
II	Experimental Data With Blowing	7
III	Types of Unblown Flow Data	10
IV	Types of Blown Flow Data	11
v	Test Conditions (Nominal) for BLIMP Comparison Data Sets	16
VI	Description of Instrumentation	17
VII	Determination of Key Parameters and Methods of Data Reduction	18
VIII	Magnitudes of Momentum Integral Equation Terms	92
IX	C _f /2 Comparisons for Stanford Runs	135
х	Variations in Momentum Integral Terms for Jeromin Comparisons	149
XI	Typical \overline{n} Distributions in the Outer Wake Region	173

فقرأ المعرك معميرين

والمراكبة أتحرك

د از این مربق این مربقه مربقه میرد این این میرد میرد. مربقه میرد این مربقه این مربقه میرد این مربقه این مربقه این مربقه مربقه می

ى بىرى بەربىلەر بەربىلەر بىرىيەر يېچىنىيە يېچىلەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرى بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىرىيەر بىر

tunidity and on choses

abdreisessele en a terre

and the second second

wyie jadah lif benel

z

STREALS

λ	parameter used in Cebec: mixing length formulation, defined by Equation (14); also, a similar parameter is used for the Bushnell and Backwith formulation, defined by equation (25)
λ.	perameter used in Soshnell and Beckwith mixing length formulation. found from graph in figure 9.
c _f	drag coefficient; $\sqrt{\frac{1}{2}} c_1 u_1^2$
C _{fi}	drag coefficient in an incompressible flow
с _р	frozen specific heat of the gas mixture
F	dimensionless wall mass flur per unit area, $(vv)_{v}/(cv)_{e}$
h	static enthelpy of the gas
* h	reference esthalpy, defined by equation (33)
e,	incompressible form factor, $s_{i}^{*}/\varepsilon_{j}$
FT	totai enthalpy
k _m	momentum aiking length constant, 0.40-0.44
k _h	energy mixing length constant, 0.44
K.	mass fraction of molecular species i
L	mixing length, defined by equation (4)
M	Nach auster
N	number of modal points across the boundary layer selected for the purpose of the numerical solution procedure; also parameter used in Cebeci mixing length formulation, defined by equation (15)
P	pressure
P ⁺	parameter used in Cebeci mixing length formulation, defined by equation (16).
Pr	frozen Prandtl number of the gas mixture
Prt	turbulant Prandtl number, defined by equation (11)
Pr _T	turbulent Prandtl number based on total enthalpy, defined by equation (27)
đ	diffusion heat flux per unit area
τ	recovery factor
r _o	local radius of body in a meridian plane for an axisymmetric shape

Xİİİ

SYMBCLS

(continued)

<u>Re</u>	Reynolds number; subscripted with the length scale if other than \mathfrak{s}
sc _t	turbulent Schmidt number, defined by equation (12)
2	tesperature
u	velocity component paralllel to body surface
v	velocity component normal to body surface
x	distance along the body surface
Y	distance from surface into the boundary layer, measured normal to the surface
y _a	constant in the mixing length differential equation (see equation 5)
ř	boundary layer outer edge intermittency, defined by equation (19)
:	boundary layer thickness
ŧ	displacement thickness
* * * *	incompressible or velocity displacement thickness, defined by equation (8)
ćđ	average turbulent eddy diffusivity for all species
ⁱ đ _i	turbulent eddy diffusivity for species i, defined by equation (10)
د <u>ہ</u>	turbulent eddy conductivity, defined by equation (9)
Ê.	turculent eddy viscosity, defined by equation (3)
-	transforzed coordinate in a direction normal to the surface, defined by equation (41)
ê	ropentum thickness
÷	incompressible momentum thickness, defined by equation (26)
5	shear viscosity
-	kinematic viscosity
•	transformed streamwise coordinate, defined by equation (40)
-	density
ి. వి మామ్	totei mass flux per unit area into the boundary layer
:	local shear stress

SYMBOLS

(concluded)

Subscripts

- aw pertains to adiabatic wall
- e pertains to boundary-layer edge
- i pertains to the ith species or to the ith nodal point in the boundary layer, starting with i = 1 at the surface
- n pertains to the nth nodal point, corresponding to the owter edge of the boundary layer solution
- r pertains to recovery or adiabatic wall conditions
- w pertains to wall
- 0 pertains to stagnation condition
- 1 reference condition, usually taken as zero streamline from inviscid solution (synonymous with boundary-layer edge in the absence of an entropy layer)
- ∞ pertains to freestream

Superscripts

- κ equal to unity for axisymmetric bodies and zero for ιwo-dimensional bodies
- () evaluated at the reference enthalpy condition
- () signifies a turbulent fluctuation
- () signifies a time-averaged value

SECTION I

INTRODUCTION

The Air Force Weapons Laboratory recently sponsored the development of boundary layer computer code (reference 1) for use in predicting multicomponent, chemically-reacting, laminar or turbulent flows over ablating, reentry vehicle surfaces. This code has been designated BLIMP for Boundary Layer Integral Matrix Procedure. The turbulent model used in this code was recognized to be relatively untested for the high Mach number boundary layers of interest to the Air Force, although it had been shown to be satisfactory for low speed unblown and blown turbulent boundary layers (reference 2). This report presents the results of a study to validate and to improve that turbulent model for compressible and nonsimilar flow cases.

In this study, the available turbulent boundary layer literature was searched in detail for experimental data of sufficient accuracy for detailed boundary layer profile and integral parameter comparisons. Interviews with principal investigators were conducted to learn of new data or unreported data reduction techniques which are often missing from the usual final report. Five data sets were selected from approximately one hundred experiments originally considered.

The turbulent model in the BLIMP code was also re-evaluated in view of the numerous papers and reports on turbulence modeling currently becoming available in the literature. Limited changes in the existing BLIMP turbulent model (as described in reference 1) were made before the detailed data comparisons presented herein were finally generated. Extensive use of the BLIMP code on the selected test cases then resulted in approximately 75 graphical comparisons of the code predictions with the experimental data. Questions of code starting procedure, nodal spacing, etc. were also addressed in order to give a complete picture of the use of the BLIMP code and the kind of accuracy that can be expected from it.

This report discusses the turbulent model study in essentially the order in which it was carried out. Section II describes the literature survey phase including the criteria used for selecting the final five data sets. Section III describes the turbulent model used in BLIMP, and presents some limited comparisons with other turbulent models. Section IV presents the comparisons

between BLIMP predictions and the experimental data. Section V contains some conclusions and recommendations for further work. A general discussion on the use of the code for turbulent flow problems is included in an appendix.

SECTION II

COLLECTION AND EVALUATION OF TURBULENT BOUNDARY LAYER DATA

The number of experimental studies of turbulent boundary lave: flows which have been conducted since the turn of the century is possibly in the tens of thousands. Of these numerous studies, possibly 1 to 10 percent were carried out with sufficient accuracy and detail to yield data of "acceptable" quality. Of these high quality experiments, perhaps 10 percent include the type of flow configurations and the necessary instrumentation to provide data of interest in a turbulent model study. This leaves possibly a few hundred papers and reports which should be reviewed for a study of the type being reported here. This survey was limited (by the constraints of both time and practicality) to reports or papers generally found or referenced in the recent open literature. Thus, approximately one hundred documents were surveyed briefly for pertinent data. Descriptions of test configurations, test conditions, instrumentation, and the type and quality of data taken for the most relevant experiments is presented later in this report.

In the remainder of this section of the report, the experiments referred to above are summarized and discussed in Section II.1. Criteria for selecting the best data sets are presented in Section II.2, and the final selected data sets are discussed in some detail in Section II.3.

1. SUMMARY OF EXPERIMENTAL DATA CONSIDERED

There are several ways in which the various experiments can be described; however, it is felt that a presentation in tabular form is the most efficient and will be most useful for future reference. Thus, two tables are presented here to summarize the experimental data which were considered. Table I includes flows without blowing, while Table II includes flows with blowing. Within each table, the experiments are placed in alphabetical order, according to the principal author's name in order to facilitate cross-referencing with Tables III or IV or the text of this report. A shorthand notation has been introduced for some columns (SFB for skin friction balance, etc.) and a "Remarks" column has been included to give additional details about the experiment. Often the remarks refer to the suitability of the data for turbulent model studies, since it is difficult to glean this information from the other tabular data.

					CONTRACTOR OF THE						
Author	Configuration	£	1 1 1	<u>ب</u> ،	تر م	11	67/44	.1.1.2	Profile Internation	1010-00101 Intransion	
hároch. el el. ¹ 1465	N. Lanity Mar.	•;	•	к.н	541	1	8	:	titet perm. Nitet tang pron. Latet tang pron. Kalilatan pantan	ta tare 17 a inde uit of stinder	bela anter well ant con- statement verh well about
Allon, ⁴ 1378	and then bely of resulting in the	-153	, , , , , , , , , , , , , , , , , , ,	:	3	1		:	P. protes at N -	!	
brinich & Musican, ⁵ 1979	Necessities in the contract of the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on the contract on	. ,	ļ	•	ĩ	1	The state	1		ł	thethercel acte unly. MGA TH is propulation
Nott, N. 41.		3.4-4.6	0.77_3.5 # 20 ⁴	1.4-10.4	ž	Ł	i		Pa Firme, Pa Probe squilibries long probe. firm uits Tick probe	i i	A continuation of the Loo ot. al. intestigation to pressure gradient fimme
Carry, 7 IM4	Compared flats plats in Linedian 20" by December 14" mol	33		:	Ŧ	Ł	1	• * • • • • •	1	T/C m bechaids of plate to men but transfor	minimized of all view
Colon, ⁰ 1951	Flat plate in JPC supersonic turnel	2.0.2.4	100	4.24- 1.15	350	t	1	ţ	.		Encellent of empori- mate
tacultor."	Low speed temes! with blowing up- titions of best ascrime	£0.0-	8 a. a. 8	-1.	3	1	t	:	1	r. •	transitiont jou spaced duta survey included
Fischer et al. ¹⁰ 1978		x	0.71518.9 81 #	ï	Ŧ	\$	I	ŧ	P. Probe. P. Probe.	J	Mormal pressure seation. La boundary layer
Gradwr et al. ¹¹ 1967	5 mil ands com in two different short tunnels founded and come mustion	Le fee con- le seculo 3-15 fee concerté marite	9 4 19 <mark>9</mark>	ie.	2500. 4154.	£	yet with com- tour mottle	• • • • •	t _o prote	r, tope, platt- mer this film page	buta presented in 1974. Liciari detail fur mudul studion
4111. ¹²	There as hyper-	•	2.5 a 10 ⁶	9		£	ting gradiant	•.5•	ij	14 cm	Data die yuite scitterad
Mophime & Mercen,13 1948	March tune flow tates a cylimet three and is discont and is	5.(-1.5	-1.0 x 146	H	itelle tertise	£	1]	Platiane colo- rimeters at 5 statione	Calactimeter data gurattur- able starw pages protrumed tata flow
Jame & Puller, 14 1970	Langley 20° hyper- sould tunnel. Theis on south well	3.9-3.6	0.5-1.13 81 %			Ī	-	•••	7. 7.1.4. 1. 7.4.0	and the second	cold please but. May attact data
Julion of al. ¹⁵	"tanford for speed beet and muss trans- for appurates	0.02-0-10	2.5-3-4 # 10	-1.0	594.	1	stant, when by 4 acceler- iting floor	:	he protec	T/C is will	faily whenly dues in this rights
Kealmey at al 16 1978	Stanford Jou speed hert and muss trans- for apparatum	and and a	4.7 × 16 ⁵	••••	.910.	£	ites ler at 1 mg	1.00			This report concentrates on heat transfer walkets
1949 st al. ¹⁷	Lower will of build morello and extension. 12 vide. wher cooled	•	8-56-6-3 # 18 ⁶	1.9-10.0	.0 -1210.	£	1	•.5-1.6	P. Mah. Plat Prat. Prat.	STR (verser cool w) T/C's in insul.cue for heat itensfer. P _R tues	Appears to be a year with of equationals. All date in tubular form
laua la	AIDC tunnel m. 4. 20° dien, conjed will cylinder aligned vith flow, Reservents along imide will.	¢, ¢	6.1 # 1- ⁶	92.8	Ę	£	2	1.24-0.16 2100 Mill		ra Lapa T/C'a at surfact Bass flue calori- Veter	Tangerature profiles in built that a loss time to reach equilibrian ufter the time. Cand then the computitions.

NOT REPRODUCIBLE

AFWL-TR-71-57

-

These I femalesed

Ref or 1. ¹ Manual manual manual stands 11 and 10				יי צי	• 1	•	7	ž	Ī	Profile Lotrementation		
Rest of the second state and state state Letter between the state	1. · · · · ·	Stations for speed	A THE R	4.7 2 10 ³	:	3	£		7	11 12 2.7		
American Employed method International method Entrop method International	R. 7 7 7 8		5.0-1.2	*. N-4. 275.10 ⁴	3.11-31.6	911-645	Ł	Ĩ	6.6-4.4	1		
Internet: Description Description Description Description Internet: Internet: Internet: Internet: Internet: Internet: Internet: Internet: Internet:	time + Trimer, W		•••	•	6: 9-54 14-132		E	ł	6.2-5-2	ļ []		
martened production location location location location location location location location location location location location <	11 11 11	l'uld" drown blan- dron vied tambel. Totto do will	13 51 22032	1.0 H 10 1.0 H 10 1.0 H 10		Ŧ	1	1	:	**************************************	Sa Patron Trinita Mili	Profile and Annual Lenne after Line
No. or 1.1 ¹ Terrene were branching in the second of th		The Place is "	1.7-4.4	6.6-9, Julgh	í .	ł	2	1	ł		1/2's (s w)1 7, 44	ten ift mensues. was in
A to define the state of th	m. L. et	transfer survey			•	ź	1	Ł	ł		***	bets in regulated for any
r 1 manuelle Universitie 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1	lat and Kaya, 25			2.6 x 10 ⁵	<u>*</u>	38.	Ĩ	2	8.87-1.65	Į	TAC's in plats	This is the first reserve
<pre>munta a tail is interested in a state a s</pre>	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	Le annale ta	:	¹ 01m(0.1-51.0	3. 32		t	1	1	P. Prada cale (castinuova tre- versi)	ŀŀ	the liture approaches
¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹ ¹		le ⁿ tectudes antis the in short travel]]]		11-11		Ŧ	£	•.14•			Mails desim your silver rays
The mark hand, if the first mark hand, it is a set of the first mark hand. It is a set of the first mark hand, it is a set of the first mark hand, it is a set of the first mark hand. It is a set of the first mark hand, it is a set of the first mark hand hand. It is a set of the first mark hand, it is a set of the first mark hand. It is a set of the first mark hand, it is a set of the first mark hand. It is a set of the first mark hand hand. It is a set of the first mark hand hand hand hand hand hand hand hand	1			0.61-4.9m106	Ŧ	Í	1	1	1.1-1.2	Reut 1. bang. probe	TA THE ACTION	Busils and reach they well. Busils and reach the well-
Y and factors the many set of a set of the life life life life life life life lif	* ***	MK 1.3' about two- mai in turbulant durt configuration	1.0 Ithreet Ferlicet	, • r • ,	, , , , ,		£	1	· . •27	I	Mail celorimener. F _e menored with	Tradiers with the reviews.
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Numerous an and the units of Separ-	5-11-8	1.9 = 10 ⁶ 7 1.01 = 10 ⁷		1448-1986	Ł	L	N.9-X.9	Mak speed transs.	ctystal pages Thin film heat transfer upp	Punaley time (s-
Huild Muild will beste it. A why is build has-bee yes yes culled. Some parameter the second of the s	tis et al. Jl	Live as Adread. et al. vith well continue	;	,	15.7	E	ŧ	8	9.4	trades	Meat thow metaru: 2 T/C's on Cunti	This is a constanuation of the shirter, of a marries
ide of al, ¹³ Standown low speed 0.44 2.4 x 18 ⁵ 1.6 3.14 seedl no 0.97-1.531704 Tria 1.01-0 m	•• 32	Maile will track At 2011. The murity	•	÷ 3 3.5	2,41-65	8992-5997	Ē	E	·.n-a.s	Smic promotic probe for r _o 4 r _o .	repett	Data scatterrad
		Stanford Ion speed inst and Nuc trad- for apparatus	5-	2.6 = 10 ⁵	1.6	9	1177	1	1.97-2.65	Larye dimastona" •	TT's in plate I. Tape	That follows to be mode an December back models
annu at al. Station law peerd 0.: 7.4 16 ² 1.1 5.4 Tee yes 3.31-1.65 8 probe 8 poots 27 Not appendix for appendix		stuntend low speed best and mass trang- ter appuratue	9 11 4 2 4	2°6 = 102	•	BES	E	Ē	1.95-1.05	a, prote	es posts 177's in plate	und rays This report continue tour persiver data, tou in con- junction with Julien reput.

÷

TARKE I (concluded)

hearts	Data is scattared	How Lastrematation tothiges for denaity measuremants. Deca fairly scattered	This temptative and to me used in conjunction with Lingson velocity data	This is considered to apode a classic set of low apode data	Couled model becad up were a file	Unintensional Verticies in 2. and Ta betware twos and alle reduction difficult
The trumments then	Pa porta thin fiin re- sistenes tyr's, ###	rus (la ber	ryk'a in place	ł -	wit tr's w " se chin chin calerimera	Ņ
Profile Latreastation	r.a.t.e	Parate the for for the second the		Pa pruče : aka	Į	H
5	u.16-0.34	1.64-2.78	1.1-1-10.0	!	• • • • •	0.66-2.0*
#2/4#	11	1	1	1	1	1
11	Ł	Ł	L	1	L	£
₽° •	H(11]	\$195-9662	83.	• 518	11-11	11-17
••	-99(T-9"9(N-91	•11 -	• 1 •	1 1	6.6-7.5
172	•••1=1••	1.1-2.14105	2.6 # 20 ⁵	6.6 ± 10 ⁵	91 = 0.11-5.0	3.9 ± 10 ⁶
=	1.11.7	4.1-1.5	H'H -	4.013	•:	
rcalignration	Flat plate in about turned plue mar- surgements on massle	These an anoste unil to aboot transal	Stanford low append heat and mass trans- for appendius	b) Wooden plate in low speed wind tunnel	NOL hyperconic tur- mil. 10° charp ocea model vich actural treasition	NOL 12 m 12 cm by hypersonic tummel, tests is boundary leyer
Author	11.2.13	111.000 ³⁴	11 m 13	separte è Tillann. ¹ 14	11 mont 1	intler 4 Che. 40 159

Hotoni Hotoni Post virg amenon Post virgention trap Post virgention P

T./T.m. ratios

NOT REPRODUCIBLE

uther											
	Configuration	F	₹ ⁷ 2	<u>ب</u> 1	۴° ه	In jectant	llaw § P	. <u>.</u> 5	rrafile Instrumentation	Mail Instrumentation	
	Lo' speed 7"s7"mi" U.of Utah temmel with blowing on one wall	(20.0	1.54 x 10 ⁵	-1.0	8	•15	Ł	STR. 11.1	Pe probe, hot film probe, thermistor orobe,	T/C'L Schedard and on surface	Good study of tembulent fluctua- tion offects. Theuficiant dis-
	Low peed jet impinging on ramspired wall	3. 3.	2.03 # 10 ⁶ 9 x * 0	-1.0	975	air	2	1	Po probe take	and the second se	custion of plate uniformity
t) (1)	<pre>Jy = 4 * section of twa- mai val for a percentic remediar USC and CD Con- vair. Their place van 5y = 16*</pre>	o	4.1-7.5 x 10 ⁶	1.14	\$6 9		ţ.	•	Some use of probet. but deta mot ra- ported	7/C's at surface P ₂ thps Quard hanters	Good stady of sffects of various injections an will heat treasfer
_	MOL hypersuchic tunnel Bo. 4. Blens plate mented in tunnel	5.1	4.6 × 10 ⁶	9. e	ī		9.	4 6.9	Po probe	T/C's in top surface of plate	Poor injection cass flux dis- tribution
_	Flat plate mounted at exit of 2-D monstle	•	2.4-5.6 ± 10 ⁶	15.2-30.0	064		yes	5.1	P _o probe Equilibrium tem- perature probe		Transition over significant portion of plate
¥.	Poroue well im 3.5" m 6" wind tunnel		6.9 6 13.8 x 3 ⁶	3.7 6 7.6	560-585	*	¥	1	Pitot mesurements rade but not pre- sented. Also some to meas.	Porous SP	ŝtrictly C ₈ Jata
	Porous wall with side fences in 2° x 24° wind tunned	3.4	2.6 × 10 ⁶	0.1	240	* i c	¥	I	Fe probe	T/C's on backwall. F ₅ Measured on sidemails	Concentrated on string blowing where C _f · · d-/dx
	Small 2-D half twn- me! with porces test section in flat wall	2.5 4 3.5	1.43-1.52 ± 10 ⁷	5,2 4 8,3	522	a i e	Į	-:	P _o proba, equilib- rium	7/C's mean surface. 1/C's to get qu.	Good discussion of plate uniformity. Tabalar date giween by Équires.
	Same as Alsaji but new porcess plates	50.e	201 × 1.2	D. [-	-519 bvt combus- tion	#2, H2	.	1.10-	P. probe, P. probe, tempera- ture probe	In depth and bect- face T/C's	Combuscion of B2 in Downdary Layer. Looked at both mano a negative pressure gradient. S0 discussion on plate waifors.

TIMUE II

ł

					TIENT .	E II (contant	ĩ				
Author	Configuration	×	2 ¹ 2	~ ⁵	» «	Injectant	11	- 1/2	Profile Intrumatition	ttal I Tast running to D	HEALS.
Julien et el. ¹⁵ 1969	Stanford low speed heat and mass transfor appa- satus. Pertanglar Froom- section duct - 20° vide. F high. 24 segment percous walla.	0.023- 6.107	201 x C.1-7.1	8 * 1,	oc\$-	ate	8	:	P. probe	7/C°s in plate. P _a tapa	This import commentates on velocity date.
Kwarney, et al,16 1910	Stanford low speed heat and mess tranifer appa- ratus	0.021- 0.064	201 x c.)-1.1	• 1· 0	915 -	i.	ţ	1.09	P. probe 19C probe	t/C'm im plate. Po tape	This report concentrates on beat transfer effects.
Kent, et al, 50 1969	AgnC tummel B, 50° dis. test section, porous vall come model, vari- ous nose bluntness ratios	a . B	0.55-3.6 ± 10 ⁶	not given	0961	°, 3	ł	;	Adjacent Fo. To. and concentration probas	Fs taps. T/C's im wall	Primary purpose of these tests was indeced force and surface presents data, profile measurements are of poor quality.
Kulqein, Sl 1962	Porous tube with founded momepters allymed with low speed flow in 1% ft. square tumbel	800-9	5.5 x 10 ⁴	-1.8	0C S -	ย์	ŗ	2.8+ 2.8	P _o probe, T/C probe, conceptra- tiom probe	T/C on wall, used P _o probe as Preston tube	Date in graphical form only. Reynolds number looks very low for completely turbu- lent flow
Martallucci ⁵² 1910	Sharp come porous model is ALDC Twamel B, 18" base x 48" long model	:	0.30-3.8 x 10 ⁶	41.1 31.1	o \$C1	ž	8 7Å	1.54- 9.14 Varied Along body	r _o probe T _o probe	P ₄ taus T/C's (one in each porous wall section)	Similar to Reat, et al. but much better documentation. Still trouble with variations in blowing fate
McQuaid. 51 967	Low speed tunnel with 28° x 40° porcess will section fixyle com- partment for transpi- ration specem	0.136 0.136	3.0-9.2 x 10 ⁵	-1. 0	9 75	aí t	2	Ì	Po probe	P. Lape	Ome argetiment with discom- tinuous injection, but poor data
Moffat & Kaym, ²⁵ 1967	Stanford low speed heat and mass transfer appa- rates	10.0	2.6 x 10 ⁵	0.1-	• 5 3 0	ł	11	0.97- 1.05	P.o prote	T/C'a im plate. P. tapm	Excellent experimental facility
Muzay, 54 1967	Porous were in 7" x 7" for speed tunnel	650.0 850.0	2.16 4 4.0 x 10 ⁵	o.1.	962-	*	2	ł	Pitot tube	tion	Continuetion of Monidrige and Musry work
Pappas 6 Churo, 55 1960	Ames 2' x 2' transonic tworel 6 10' hear trans- fer twomel. Come rodel of molded fiber met.	0.1,0.7. 3.5,4.7 (freet	1.01-6.75 = 10 ⁶	not jiven	not given	eir Be cc1 ₂ r ₂	2	I	Norm	P _a tapa	Meaved total drag only with model stime stain gage. None injectant distribution

NOT REPRODUCIBLE

·

these II (concluded)

.

Anthe	Configuration		2	- °	••	lajactaat	4 (mll	5	Profile Increased	m11	
				ş							
Mana & diamond	Partone surface sharp cons		1.0-0.9 × 10 ⁶ (mm comme)		ġġş	ja Brit	•	Į	tione -	surface T/C's	fair anomat of scattar from ray to ray. Buta in graphi- cel form only
Ntaroom, et al. ⁵⁷ 1964	20° Mach (Langlay two- mai. Maray indiang odda (13) pictu madia vice pictus auctiona of slot. Piata wu 24° lang x 17° vite	•	1 9	a.1	£	alt	1	I	P. 1004	578 () x = 20°,), tape	looked at effects of steps is blowing, mittyle slots, etc. Bets is graphical (ees
And a second sec	Low speed turned with pores will section	-540-9	4 14 5.5 = 19 ⁵	0-1-	818-394	uir. m. 03. cl ₂ r2	L	1.0-	A protection	T/C's on surface and backside	Journal article does per give profile data. Orivital report la Mussian
limon, st al, ³³ 947	Stanford lor mead heat and must treatfor appa- rates		2.6 = 10 ⁵	•1.6	965-	air	T.	1.1	ante de	Prints in place	Took data vich Moffat. Excelient experiments.
11. 12 141.	7° m 7° m 8° 0, of Utah Jow speed tumbal	629-9	1.55 = 20 ⁵	•-1.0	915-	M2, M2 alt	1	1.01	Po. 1/C, essent- tration probes, bot film probe	T/C's on surface and imbedded	This thesis concretates on bot film development
andro. ¹⁰	from an Jurnala, Mali- tian injectant bart transfor 6 transi drav- down capability	12	0.9 ± 10 ⁷	1.62	*		1	1.05*	Pitot t ubu	Laws as Jeronia	This report also presents Jeronis's H = J.3 dats, and extends study to other Mach mediers.
Mieltahr,et al, ³⁴ 1969	Stanford Jor speed hast and mass transfor appr- rates	ij.	1.55-7.66 x 10 ⁵	•-1-	R,	ale	Ł	12	2. probe 1/C probe	F)C's in plate	Acceleration flow, blowing and section
Mittan, et al, 17 1967	Staafard lov gaad baat and man transfer uppe- rates		1.3 4 2.6 a 16 ⁵	-1.0	ers-		ł	-16.0	Percent	Sector parts	Daifern velocity, with steps in will temp and blowing retes
Haberly, et al 61 1970	Appersatio are basted temmel. 7.390 percen- cens vita halo in aver- compartemented injection	żż	1.0-4.0 x 10 ⁶	ant gives	anti stan		ĩ	1	1	T/C's 6 caleria- eter in will	Film injection at more complicates results.
Similar, et al, 62 1969	Arc-tunnel in zerronaia duct configuration. Nalating teflom will in trunnel.	1	not given	n. e m.	÷.	taflam (ablation)	Ł	1	P. probe at duct stit	TVC's (proof deta).	Most of data from turbulent tests published in another report, NOL TR 69-122
boldtige 6 Muiry ⁶) 1966 Maei	7° x 7° duct with porous will		2.0 x 10 ⁵	-1.0	8 5,	, 1	(confine tion)	1	M. (some appert- mental, conditation P.T. concentration tanging probe, P.	P Lape, T/C mar electore	Measurements at one station only
Starrad (') quant MM = bot vire a Pot vi Pot vire a Pot vire a Pot vire a Pot vire a Pot	tition are T _o /T _m ration memory presents transferration transferration transferration transferration										2 11

.

NOT REPRODUCIBLE

AFNE-28-71-57

فلاقه والمتلا المتحد والملا

Mhile the information provided in Tables I and II is useful for detailed review, it does not provide an overall impression of the availability of certain types of experimental data without a detailed review by the reader. Therefore, Tables III and IV were prepared. Table III applies to unblown flows while Table IV applies to blown flows. Referring to Table III, Unblown Flows, it is seen that very few subsonic cases are listed. This is entirely attributable to the fact that subsonic unblown data were not sought in the literature search. Matting, et al. (reference 22) plus several reports of the Stanford series present data on subsonic unblown layers; therefore this category was included for completeness. The other entry is from the 1968 AFOSR-Stanford-IFP Turbulent Boundary Layer Conference Proceeding (reference 64), and is regarded as top guality data useful for reference purposes.

There is a noticeable difference between the number of reports on supersonic flat plate flows (1 < N < 4); and hypersonic flows (M > 4); hypersonic flows seem to have received more attention from the experimentalist. This is attributable to the need over the last decade for data suitable for reentry calculations and comparisons. Thus, major test facilities around the country have been constructed and used primarily in the hypersonic flight range.

Data on allelerating boundary layers were also gathered during the litersture survey place. Pressure gradient investigations are often more unavoidable than intentional in many test facilities, thereby yielding rather uncertain flow data. However, there have been some research programs in which pressure gradient was varied systematicially.

Turbulent boundary layers with a nonreactive transpirant have been investigated for many years as a possible technique for heat transfer alleviation in turbine blades and reentry vehicle nose tips, among other applications. Air has been the most popular injectant for experiments, although possibly not the nest practical in many actual coeffing systems. The literature review for subsonic blown flows was intended to supplement that given in meference 2. Thus the Massachisetts Institute of Technology (MIT) work (i.e., references 65 and 66) was not included for re-evaluation. There is no shortage of new low speed dath available. The Stanford investigations have resulted in at least 13 separate reports on blown turbulent boundary layers, with heat transfer and pressure gradient effects documented as well. Strangely enough, the great interest in low speed blowing investigations has not carried over to the supersonic and hypersonic flow regions. Particularly at Mach numbers greater than four, there is a real need for further basic flow data suitable for turbulent model investigation and development. The picture is equally

TABLE III TYPES OF UNBLOWN FLOW DATA

States and states and states and states and states and states and states and states and states and states and s

entremente velhand konneditionk sus

.

.

11

AFWL-TR-71-57

TABLE IV

TYPES OF BLOWN FLOW DATA

			Nonreactiv	e Blowing			Reac	tive Blow	бu
	×	ir or N ₂ Injecti	uo	Fore	sign Gas Inject	ion			
	Subsonic	Supersonic	Hypersonic	Subsonic	Supersonic	Hypersonic	Subsonic		Supersonic
		1 < M < 4	M > 4		1 < M < 4	M × 4		,	1 < M < 4
	Alsaji	Bartle ^e Leadon ⁴ 3	Danberg ^{.4} ć,45	Pappa ^s 55,56 Okunc 55,56	Bartle & Leadon]3	Kent, et al ⁵⁰	Jones-49		Winkler et al62
	Baker ⁴ 2	Dershin et alif	Kent et al ⁵⁰	Romanenko & Kharchenko ⁵ 8	Fernandez £ Zukoski47	Pappas 6 Okuno55,56	Kulgein ⁵ 1		
	Jones 49	Fernardez 5 Zukoski 4 7	Martellucci ⁵²		Pappas & Okun55,56		Romanenko 6 Kharchenko		58
12	McQuaid ⁵³	Jeromin ⁴ 9	Pappas & Okuno 55,56				Smith ⁵⁹		
2	Muzzy 54	Pappas £ Okuno 55,56					Wooldridge £ Muzzy63		
	Pappas 55,56 Okuno: 55,56	Peterson et al57							
	Romanenko 58 Kharchenkc ⁵⁸								
	Simpson et al57								
	Smith59								
	Whitten et al 37								
	Wooldrige 4. Muzzy.43								

AFWL-TR-71-57

incomplete in the nonreactive foreign gas injection category, with Bartle and Leadon's study (reference 43), where no profile information was taken, being perhaps the most exhaustive.

Finally, the category of most interest for heat shield ablation or transpiration applications is the reactive blowing data. Work is just getting underway in low speed flows with the University of Utah duplication (references 41, 49, and 59) of the original Wooldrige and Muzzy experiments (reference 60). There is little basic data of interest in supersonic or hypersonic flows.

2. CRITERIA USED FOR DATA SELECTION

The criteria used for judging the usefulness of experimental data must of course depend on what use is to be made of the data. For turbulent model studies, it is of interest to model the details of the flow through the entire boundary laver, therefore profiles of the important boundary layer variables become of primary importance. In the usual experiment, state-of-the-art instrumentation is used to measure stagnation pressure and stagnation temperature behind a normal shock (created by the probe itself) for supersonic flows, although for adiabatic wall conditions local total temperature can be assumed equal to the reservoir temperature with only a small error. These pressure and temperature data are used to arrive at other more fundamental variables such as velocity and temperature or enthalpy. Thus, data selected for these turbulent model studies must at least include measured profiles of one or more of the important boundary layer variables.

Other criteria on which to judge the data are more subtle and can only be described in a fairly general way. Herein is presented a list of items that have been considered in viewing a set of experimental data.

- The wind tunnel should provide a shock-free, low freestream turbulence isentopic flow.
- The boundary layer flow upstream of the test section should be well characterized if it eventually forms the boundary layer for testing (as in wind tunnel wall boundary layer testing).
- The test section or model should be of known flatness and hydraulically smooth.
- Leading or trailing edge effects should be examined to determine their influence.
- The two-dimensionality of the model wind tunnel test section combination should be considered.

- The flow should be fully turbulent at the test section, free of any transitional or boundary layer trip effects.
- Where transpiration effects are investigated, the injection must be uniform or known to within a few percent.
- The displacement effects of the boundary layer flow on the inviscid flow must be completely known particularly with injection.
- Probe sizes must be small compared to boundary layer sizes.
- Disturbance of the flow at the wall by wall instrumentation must be held to a minimum.
- The data should appear relatively smooth and consistent with a minimum of scatter.
- Where possible, the integral conservation equations should be checked and satisfied experimentally to a reasonable percent error.

The points mentioned above are essentially the desired features of an acceptable boundary layer experiment. No attempt was made, however, to develop a rating system for the various data sets. Rather, an overall impression gathered from reading about each experiment was formed, notes were made regarding each data set where necessary, and in many cases, telephone or personal interviews were held with principal investigators. Important features of each data set have been noted in the "Remarks" column of Tables I and II.

3. SELECTED DATA SETS

A minimum of five data sets was desired for turbulent model comparisons. The approach used in selecting these data sets was an attempt to provide data which test the turbulent model in a variety of flow situations which are encountered in flight. This obviously means that comparisons could not be carried out for each category of Tables III and IV since there are more than five categories containing good data. The five data sets selected were:

- Coles' JPL experiments (reference 3 Supersonic flow with zero pressure gradient.
- NOL experiments by Lee. et al. (reference 17) and Brott, et al. (reference 6) - Hypersonic flow with and without pressure gradient.

- TRW experiments run at AEDC (reference 18) Hypersonic flow with highly nonsimilar wall conditions.
- Stanford experiments by Simpson, et al. (reference 33) Low speed flow with nonreactive blowing, including steps in blowing.
- 5. Cambridge experiments by Jeromin (reference 48) Supersonic flow with nonreactive blowing.

Detail: on the five selected data sets are presented in the remainder of this section. The descriptions below are supplemented by three more tabular comparisons. Table V presents the particular test conditions for each test run used for comparisons. Table VI presents further information on the instrumentation used in each of the five data sets, while Table VII compares the data reduction techniques.

a. Coles - Supersonic Flow with Zero Pressure Gradient

Coles' data (reference ?) at two Mach numbers, M = 2.6 and M = 3.7, were selected as baseline cases. Coles' experiments were carried out in the Jet Propulsion Laboratory 29-inch supersonic wind tunnel, whose flexible nozzle walls were adjusted to give a nearly constant pressure flow over the adiabatic flat plate model. The test model consisted of a 30-inch long flat plate sharpened with a half-wedge on the upper surface at each end; the lower test surface was flat over the entire length. This plate was inserted into the tunnel, located on the centerline and pitched 0.17 degrees nose down. Various types of boundary layer trips were employed by Coles. For the two comparison runs cited herein, the trip, or fence, consisted of 0.014-inch diameter wires spaced 1/4 inch apart which projected about 0.1 inch beyond the leading edge of the plate.

This experiment emphasized accurate evaluation of the skin friction coefficient with a floating element gage. Only total pressure profiles were measured; total temperature was assumed to be constant to determine velocity profiles across the boundary layer for the adiabatic plate, as described in Tables VI and VII. Preestream conditions were assumed based on isontropic expansion of a $\gamma = 1.4$ fluid. Profile measurements were taken at only one station 21.5 inches from the leading edge. Skin friction gages were placed at 5.5, 13 and 24 inches. Static pressure taps in the plate were used to determine axial pressure variations.

TAPLE V

TEST CONDITIONS (NOMINA FOR BLIMP COMPARISON DATA SETS

	Stagna Condit	tion		ь łge Co r	ditions		PM	11 Condit	ions
lest	P _o (atm)	т _о (°R)	χ ^υ	Pe (atm)	т _е (°R)	u _e (ft/sec)	Tw (°R)	T _k /T _{aw}	ĵu,
COLES #27. M=2.5	0.935	556	2.568	0.049	240	1950	522	1.000	0.0
#20, M=3.7	1.355	561	3.701	0.0134	150	2219	512	1.000	0.0
NOL Lee, 5 atm	5.1	784	4.70	0.0135	244	2766	520	0.74	0.0
10 atm	10.2	747	4.74	0.0241	: 36	2711	524	0.74	0.0
Brott, 5 atm	5.1	762	3.89 +4.55	0.039 +0.016	190 +1 49	2628 +2720	557	0.82	0.0
l0 atm	10.2	762	3.93 +4.56	0.073 +0.033	186 +1 4 8	2633 +2719	556	0.82	0.0
TRW Step-Up in 1.w	4.86	569	4.00	0.032	136	2280	17C +565	0.33 +1.085	0.0
Step-Down in T _W	4.89	573	4.03	0.031	T35	2295	560 +185	1.09 +0.36	0.0
STANFORD #2367, F=0.000	1.005	525	0.0375	1.004	525	42.2	525	1.000	0.0
#122366, F=0.002	1.003	545	0.0403	1.002	545	46.1	534	086.0	0.00190
#122066, F=0.004	7660.0	548	0.0402	0.996	548	46.2	532	0.970	0.00385
#121966, F=0.008	1.000	546	0.0402	0.999	546	46.0	527	0.964	0.00780
#5867, Step-Up in F	1.000	529	0.0380	666.0	529	42.6	532	1.010	+0.00400
JEROMIN #2.5-0.0	5.21	530	2.55	0.28	230	0061	514	1.035	0.0
#2.5-1.2	5.21	536	2.55	0.28	235	1905	522	1.070	0.00127
#3.6-0.0	8.28	533	3.57	0.097	149	2150	517	1.055	0.0
#3.6-2.1	8.30	527	3.52	0.105	152	2125	524	1.11	0.00211

.

AFWL-TR-71-57

IN JUNE

et e

	massiskiikiine aecs file	5	\$	22	CALIPANTE INNECTOS: 2 CLOTAL DATA SCA 2 CLOTAL LAN AN ANA SCALL ALS AN NACT, ALS L-NIC NAUTUCS FOR MESSAGE NAUTUCS FOR MESSAGE	CULTRATES POSY KC. NGARAE POSYNG. Postane data an Upvermene
	SKIA PRICTION	HUL-HUNCE FLANTER ELPERE LACE LOLAED 5.9. UT. AN N FROM LOOME DOC	"NAL SUA FUCIAN FUNITS ELECTION SUARTS ELECTION SUARTS AGL. NUE COLLO.	jas staattaa hadis. 8.seen aala kis	5	Ē
	WALL CONSTITUTES	5	"LE GRY: FIRMAND KK IF- FIRMAND KI HERCORDAR HERCORDALLS. Herl: AF KUSHE	Q GADIN IEU Inangen Gals. 16 - 0.255 Blu., 18 - 0.255 Blu., 11102	5	5
	30000401	5	CURAT REPORTS	R COMPLANDER Networks Aus Cardon Frank Cardon Frank Cardon Frank Cardon Frank	S INNE CONTRACTOR TREPROCEMENTS PER SERVICE & Der BLUN SWENC	11 coppet-constitution Differenciamilia Jac- Jacia Va coppet-co- Plusa Va Safface
	300XXVII	source and M- sources	**surt tas - 4.85* HL.	77 SIATIK TIMS LICATED Allow And Andread Th- SIATIK SUMMACL. 0.4897 F129 ICE	Static bes update on en state malls mortidis scal n mortidis scal n mortidis scal	5
a of instrationation	PUSITION	2	5	orta l'Inda Mila- Houtas Anacas de Cla subus de Cla subus de Classif Canadi (e Francis de La Cla transi tra	PALIK LOND RI- CONTRIA, WUL LUCATION RETENUED D. BLEETINCA, J. BLEETINCA, J. CONKE IN FI- D. CONKE IN FI- D. CONKE IN FI- D. FIETING & UL- D. MAKE NYE-	LIGA POIDTIOETR Attoch 70 Ne S.Dig Swits
N LATUDS IN	PAGFILE MONOTIES Tr. PEINTINE	5	TankLinklin (2015 1.000 Million Million 1.000 Million Million Molin On (1. 1.1 Million Million 1.1 Million Million 1.1 Million 1	••••••••••••••••••••••••••••••••••••••	5	Eduitization Conc. 10° Stalmerss Streat. Conc. 0.060° Bla.Af Bace utta Concidana Bace utta Concidana Concidanta Treato- Concidanta
	MILES MA	PITUR MADE CONT NE- SCRIPCI, TANNESSS TANCIA 21.5° FIDA LUDBITE EDAE	"PIUT MARE LEE, A. ALS"-AL JOY MOTE A. NO"-AL JOY MOTE A. NO"-AL JOY MOTE A. NO"-AL JOY STATIC MORE - CORE C. NULMOR. JO [®] CORE - C. OPE" B.L.	**VITH MOLE, A.AL* IF A.AL* SMISHE II- TURSHOL, STATIE PLONE CORE CALINGS	PITOT MORE 8.000° AFLINS 0.000° AFLINS 0.000° AFLINS 0.000° AFLINS	P 1101 PADAKS LUCATE) OR PAARE CONTER LINE AND & 1. TO CITHER AND & 1. TO CITHER STIPE. 0.000" KUGH
	A CONDITIONS TOPPONEN	F	- Mar II Generation		LION-CUNSTANT AN	FILE PAGE TO PO-
	FIEL STUDY	ş		TA CORRECT TO THE FORM	1111	NCHARD IT NOT LE MOR
	COMPILIANS TETPEDATURE	5	5	5	ticar-constan- sati lacan- confic	2
	NEXEMOLE (5	5	5		5
		21 15	đ	2	STARFORD B	NINOCO

"Retronues from reservoir constitues assume + - 1.4 and istrature conversion "All fours fourts un contra traveless lot "Individual traveless", locate 2.55° hystican of 4 all instrumentations "A point locate at us" , en "5, and "56 from the mozer from the "A point locate at us" , en "5, and "56 from the mozer from the

INNE VII

	TOPERATAR	5	-18 CONSTAN	NCAND N-	-H CHIST	MCDUF N-
	HUSSING	AUJHI 6 EMENTA	AUDALI CRISSIA	AUSHO DIREUM	ASSNET EVAL 10 STATIC PASSNES FLAND ON STRE MLL	ASSNED EDBAL TO Face stream statte
	nall thurster coopy least	5	LE GALT: MAL FAR HAL HAL- MAD IN TESALTD MAL. C ₄ CONVERS 418 AUMAL (AUL & THA AUMALINE ASSACS CONTIANT INTEL- LIDE (LONEY FACTOR = 0.19	KUSUG IN GARON MESS	5	5
	SLIM FERCINAN COEFICIEN	records director in flags ing flagor	HEAD RECT, IF FLMT- IS DUTTE: 429 COVID- IS DUTTE: 429 COVID- SUS 410 FOR 9 TO MULL MILL, 1900B 1900B, 1901D, 1000B 1900B, 1901D, 1000B - 3.38.	COMMEND FROM STANTING FROM KLIARIANI MACH REBRINGS STANK PRESIDE & MALL TOP- FLAMME	COPPED BY INC. IN A COPPEND AND A COPPEND AN	Converts Foot Nordshife Internet, and an ion. Sounds a sould 1 - Mende Sounds a sould a foot intersonation and 184. Landsonation lake
	ALIXTON	ALCALATED FROM MCD March and Static Tel- Loanes and En States Tela, Tercon-	comits from Local Not Numer and Ts- Feature	COMPUTE FIRM LOCAL MICH IMPRES AND MICH IDPOCIATING	COMMENT FRAM BERRILL I GAUTOR ECIER TACORDE FLUE MECORDE. E- FLUE MECORDE. E- FLUE MECORDE E FLUE MECORDE E FORM RELITION AND FORM RELITION	CUCULATE FOR LEAR HOL DEED FOR LEAR HOL DEED FOR STATC TOYODARDE, TOYON- HOL DEED FOR THE HOL TOYON FOR THE HOL TOYON FOR FOR HOL TOYON FOR THE FOR THE REAL TOYON FOR FOR THE REAL FOR FOR FOR THE REAL FOR FOR FOR THE REAL FOR FOR FOR THE REAL FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR FOR HOLD FOR FOR FOR FOR FOR FOR FOR FOR FOR FOR
CICHE AND RETAILES OF DATA NE	S Stain, Herchanne	CALCHLATES FROM INCL M SINGER AND ASSIMED INE SUITAL TEPPEDANTINE INE	HILIBRIAN TOPOADUR INTH DEVENTINGS AC CONVERTS IN INVESTS AC CONVERTS IN INVESTS AC CONVERTS CONVERTS ACTION CONVERTS ACTION INVESTIGATION CONVERTS ACTION INVESTIGATION CONVERTS ACTION CONVERTS ACTION CONVE	COMPILIE FROM LOCAL MACH NERVER AND TOTAL TEMEDATION	CLIDER TRIVIER	corr mark levelants. Dat 12015 - 15590 9 9 9,00 R n n 2,5 15,000 adf 2,5 15,000 adf 2,5 16,000 adf 2,5 18,000 adf 2,5 18,000 adf 2,5 18,0000 adf 2,5 18,000
ALYON OF IET MONE	The Month Street	ASSATCE CONSTANT ACTOSS THE SUMMA ACTOSS r>ACTOS ACTOS	EXAMPLE THE CORE CON EXAMPLE THE CORE CON FROM THE JEAL IN TRAILES - ASSAMED A TRAILES - ASSAMED A TRAILES - ASSAMED A TRAILES - ASSAMED A FROM THE TAMPLE MARKED A MARKED A MAR	TUPEINTURE PROFE	nd russes - Ie	FRUM CANFLIPPING MICH AURER AND MICH AURER AND AND O.S. TAR N - AND C. S. TAR A MICH F. MICH F. MICH F. MICH F. SEE MICHTP
N. NETWORK	NO MUCH	nvición"s pitol puch Fonula usina assure constant static puch sant	MANERALS PILOT MORE Correla asias unla static messare static messare	NAMESON'S PILOF MODE Formaa sessis wal Static Pressone	-mann manana 2789 - 952 Marana	NATERAL PITO PLAN FORMA NIN (ALD- LUTE PLE STRA SIATIC PLESSING
	Static Message	ASSING CONCINE: WILL Resemblers conc. The Pression	LEE: ASSING CONSTANT EQUAL TO VELL SIGNI K SURT: RESONCE DI- BECTLY	FEASINED BURGTLY DY Static Mode	A VUSHI CARSYN	ASAREN CINSTANT E ERML, TO FREE STREAM - Static
	THEM COMPLETIONS	ISOMOPIC OPIG- ISOMOPIC OPIG- Sanch Statution	, FROM PROFILE Lasingteurs	R FLON PHOFILE Exemplifies	3 BINCOLT IN DOC AND THEOREM	CONVERSION CONVERSION FOR FUEL FOR FOR FOR FOR FOR FOR FOR FOR FOR FOR
		ACCURACION OF A CONTRACT OF A	COPPUTE MORE W	COMMIE PRODE		
	20011000			HEARING THE PLACE	Autom Autom Autom	RUSAR BIRCO
	STACKARI ION	NLASHIE BIRGCILV	RUSSING) BURGTUY	REALING DIRECTLY	FLASHED BINGERY BY DARGH ETCR	EDMCX JF
	1231	\$ J 10)	đ	ł	1	(COLIN

.

NOTE: NAL NOT PEASARD
b. <u>Naval Ordnance Laboratory - Hypersonic Flow Without and With</u> <u>Pressure Gradient</u>

Many reports and documents were found which presented hypersonic flow data, and most of them included profile measurements. After discussions with various investigators in this field, the recent zero pressure gradient experiments at the Naval Ordnance Laboratory by Lee et al. (reference 17) and the extension to pressure gradient flows by Brott et al. (reference 6) were selected. Both of these investigations were carried out using the NOL Mach 5 boundary layer channel, which is a two-dimensional half nozzle with a flexible wall forming the nozzle. Boundary layer measurements are carried out on the opposite, flat wall, with moderate heat transfer to the wall controlled by circulating wall coolant. This stainless steel test surface is eight feet long and tapered from 12 to 13.5 inches wide. Lee's tests were carried out at four stations between 48 and 92 inches downstream of the throat, giving naturally turbulent boundary layers from two to four inches thick. Brott's measurements were at five stations between 56 and 84 inches, giving boundary layer thicknesses of 1.3 to 3.9 inches.

A wealth of instrumentation was used in these NOL investigations. Profile 'nstruments included a total pressure probe, a static pressure probe, an equilibrium temperature probe (measures local recovery temperature), and, in Brott's investigation, a fine wire thermocouple probe. Wall instrumentation included water-cooled skin friction balances upstream of each profile measurement location, and wall heat flux gages.

c. TRW - Hypersonic Flow with Step in Wall Temperature

The Arnold Engineering Development Center (AEDC) Supersonic Wind Tunnel "A" with a 40-inch by 40-inch test section provided the main air flow for the TRW experiments. This tunnel was operated in a continuous, closed-circuit mode. The nozzle used was an automatically driven, flexible-plate-type. Some of the test details are contained in an AEDC report, reference 67.

The test model was a 20.09-inch 1.5., 49.48-inch long hollow cylinder aligned with the flow to \pm 0.05 degrees in pitch and yaw about the tunnel centerline and supported by two struts attached to the downstream end. In cross section, the leading edge was a half wedge on the outer surface. The inner (test) surface was flat along its entire length with a surface finish of 100 u-inches. Seven separately controlled cooling (heating) copper coil circuits were counterwrapped on the outside surface. The 400°F step in wall temperature was made between 23.0 and 24.0 inches trom the leading edge with four of the seven circuits forward of the step. Liquid nitrogen was circulated in the cold wall region;

AFWL-TR-71-57

pre-conditioned water was used to achieve near-adiabatic conditions in the "hot" region. For the temperature step tests, the circuits in each region were manifolded together. A 0.375-inch wide by 0.006-inch thick piece of fiberglass tape, serrated at its leading edge and located on "he inner surface approximately 0.125 inches from the cylinder leading edge provided a boundary layer trip.

Profile instrumentation for these tests included a total pressure probe, static pressure probe, shielded stagnation temperature probe, and a constant current hot wire anemometer. The total temperature probe size (0.060 inch in diameter) precluded measurements very near the wall. Wall conditions were measured with static pressure transducers, surface temperature thermocouples, and Gardon heat transfer gages for wall heat flux. Skin friction was measured with 38 individual Stanton tubes.

d. <u>Stanford - Zero Pressure Gradient</u>, <u>Subsonic Flow With and</u> <u>Without Uniform Blowing</u>, and With a Step in Blowing

The heat and mass transfer apparatus at Stanford University's Mechanical Engineering Department, a facility built specifically for high acurracy subsonic boundary layer experiments with blowing and suction, was used by Simpson (reference 33) for the selected uniform blowing tests. A complete description of this apparatus is given by Moffat (reference 25). The main flow system was a continuous operation open-circuit consisting of an inlet filter, flow control valve, blower, heat exchanger, screens (for flow straightening), settling plenum chamber, primary nozzle, and test section. The rectangular test section was formed of two fixed side walls, a fixed flat bottom wall which included the porous test surface and an adjustable top wall of plexiglus controlled by a bank of jacks.

The test surface consisted of 24 individual, 1/4-inch thick, porous place sections which were 20 inches wide by 4 inches long, resulting in a total length of 95 inches. The plates were constructed of 0.005-inch sintered bronze spheres which provided a surface finish of 200 L-inches. Final assembly of the plates left a 0.032-inch gap between adjacent sections. This gap was filled and smoothed but was not porous.

The secondary, or injection, flow system consisted of an inlet filter, blower, heat exchanger, and flow header followed by 24 separate flow control valves, flow meters and delivery tubes, one set for each plate. A porous sintered bronze pre-plate was used to obtain uniform pressure in the small plenum behind each porous surface plate. Localized transpired flow rates were found to be \pm 6 percent over the center 6 inches of the span. The heat exchangers in the main and secondary systems were controlled to provide equal flow temperatures

ANEL-22-71-57

A trip of 3/P-inch wide coarse grid carborandam garmet paper was located at the emit edge of the transition section joining the possile to the test section. The trailing edge of the 1.842-inch thick trip was 1/9-inch upstream of the porces test section.

e. Jeroniz - Supersonic Flow With and Without Uniform Slowing

Jeromin's experiments were carried out in the Cambridge University Engineering Department Supersonic Wind Tennel, an intermittent blowdown tunnol with about 60 seconds of useful test time. The nounle was pro-dimensional with fixed side and top surfaces. The top surface was contoured but not adjustable: the lowar surface which contained the flush mounted test surface was flat and could be adjusted vertically and in a proch mode. The rectampular test section was 6.5 inches by 4.5 inches

Injection air was taken from the main timbel supply, passed through a flow control value and a metering prifice. Two doors carried the flow into separate plenum chambers upstream of a porous pre-plate. A single plenum on the other side was common to the entire test surface plate. Mass flow variations among 2.5-inch digmeter samples of the test surface material not the test carface itself: were \pm 5 yercent. So local variation measurements of the artical test plate were reported.

AF#1-15-71-57

Jeromia used a pitct probe and an equilibrium come recovery temperature probe (similar to the NGL design) to obtail the boundary layer profiles. After several measurements which showed excellent agreement with the static temperature relation

$$\mathbf{T} = \mathbf{T}_{\mathbf{w}} + (\mathbf{T}_{\mathbf{r}} - \mathbf{T}_{\mathbf{w}}) \left(\frac{\mathbf{u}}{\mathbf{u}_{\mathbf{e}}}\right) - (\mathbf{T}_{\mathbf{r}} - \mathbf{T}_{\mathbf{e}}) \left(\frac{\mathbf{u}}{\mathbf{u}_{\mathbf{e}}}\right)^{2}$$
(1)

total temperature profiles were not measured. Velocity and temperature profiles were determined iteratively using the above relation and the Mach number profiles. Stim friction was not measured directly, but calculated from the momentum integral equation, the Sommer and Short T' method, and a transformation method described by Jeromia (reference 48). The mass flow through the entire plate was measured with a simple calibrated crifice.

SECTION III

TURBULENT MODEL SELECTION

The BLIMP code solves the partial differential equations expressing the conservation of mass, momentum, energy, and species for a viscous fluid. In turbulent flows, the equations are written with velocity, enthalpy, and species concentrations expressed in terms of mean and fluctuating components, i.e.,

$$\mathbf{u} = \overline{\mathbf{u}} + \mathbf{u}^{\mathrm{s}} \tag{2}$$

The time averaged equations of motion then include time averaged products of fluctuating quantities such as $\overline{v^{12}}$, which distinguish the equations of motion from their laminar flow counterparts. Turbulent modeling involves the mathematical description of these new flow variables such that the number of governing equations matches the number of unknowns.

The BLIMP code has been written to accept eddy viscosity or mixing length descriptions of the turbulent fluctuation terms. The eddy viscosity approach characterizes the turbulent velocity fluctuation as being related to local velocity gradient:

$$-\overline{(\rho v)' u'} \in \rho \varepsilon_m \frac{\partial \overline{u}}{\partial y}$$
(3)

Mixing length is related to eddy viscosity by the relation

$$\varepsilon_{m} = L^{2} \frac{\partial \overline{u}}{\partial y}$$
(4)

Many eddy viscosity and/or mixing length models have been reported in the recent literature (references 68 to 74). In this section of the report three widely used models are reviewed briefly and a "best" combination of eddy viscosity and mixing length relations and associated constants is chosen for the data comparisons of Section IV.

1. DESCRIPTION OF LEADING TURBULENT MODELS

a. Aerotherm Model (reference 1)

The Aerotherm turbulent model was formulated specifically for boundary layer flows with blowing. It was first reported in reference 2, and was limited to incompressible flows. Turbulence is described in the wall region by a mixing length equation, which approaches Prandtl's expression, $l = k_m y$, far from the wall. The incompressible expression is

$$\frac{d\ell}{dy} = (k_m y - \hat{z}) \frac{\sqrt{\tau/\rho}}{y_a^+ v}$$
(5)

The constant k_m is taken as 0.44, while y_a^+ , related to a laminar sublayer thickness, is held constant at 11.823. Shear stress $\hat{\tau}$ is taken to be the local value rather than the wall stress, τ_w . This expression allows a smooth transition from zero turbulence at the wall to large turbulence in the "law of the wall" region of the boundary layer. The mixing length distribution is automatically changed for blown flows due to the blowing effect on shear stress. Constants y_a^+ and k_m remain the same with or without blowing.

For compressible flows, equation (5) was changed to deal with the products ρt and ρy :

$$\frac{d\rho t}{dy} = \left(k_{\rm M} \int_0^y \rho \, dy - \rho t\right) \frac{\sqrt{\tau/\rho}}{y_{\rm a}^+ v} \tag{6}$$

Constants y_a^+ and k_m are left at their incompressible values of 11.823 and 0.44, respectively. As with incompressible flows, mixing length distribution is changed for blown flows implicitly, by the influence of clowing on local shear stress, τ .

The equation for mixing length is solved and then used to find eddy viscosity from equation (4). This wall region eddy viscosity is used from the wall on out into the boundary layer until it exceeds the wake region eddy viscosity expression

$$\varepsilon_{\rm m} = 0.018 \ u_{\rm e} \dot{\varepsilon}_{\rm i}^{\dagger} \tag{7}$$

AFWL-TR-71-57

at which point the constant wake ε_{μ} value is adopted for the remainder of the boundary layer. In equation (7), v_{e} is the local boundary layer edge velocity and δ_{1}^{*} is the kinematic displacement thickness,

$$\delta_{1}^{*} = \int_{0}^{\infty} \left(1 - \frac{u}{u_{e}}\right) dy \qquad (5)$$

In the energy and species equations, terms involving time averaged fluctuations of enthalpy and species concentrations appear and are related to the mean flow variables by definitions of eddy conductivity and eddy diffusivity:

$$pe_{h} = -\frac{\sum_{i} \pi_{i} (pv)^{*} h_{i}^{*}}{\sum_{i} \pi_{i} \frac{\partial h_{i}}{\partial y}}$$

$$pe_{d_{1}} = -\frac{(zv)^{*} \pi_{i}^{*}}{\frac{\partial h_{i}}{\partial y}}$$
(9)

In the Aerotherm model, eddy conductivity and diffusivity are simply related to eddy viscosity by constant turbulent Prandtl and Schmidt numbers.

$$PT_{\rm L} = \frac{\delta_{\rm H}}{\delta_{\rm L}} \tag{11}$$

$$Se_{\chi} = \frac{\epsilon_{\pi}}{\epsilon_{\mu}}$$
 (12)

A value of 0.75 for each of these was recommended in reference 1.

The Aerotkern model has not had the open literature exposure that the other two models considered here have had. Therefore, in order to verify its usefulness, some results from previously run program predictions are included here. Figures 1 to 4 present comparisons with flat plate, unblown, incompressible flow data taken by Wieghardt and Tillman (reference 38). The Wieghardt and Fillman case was included in the 1968 APOSR-Starford-TFF





2

-

1000 a 11 - 11 - 11 - 11



Figure 2. Linear Loy Velocity Ratio Profile Wieghardt, Subsonic Flow, Zero Pressure Gradient

NEWL-TR-71-ST



Programme, Namentum Thicknew Keynolds Number versteewige foodston Missiburdt, Auhandru Plow, fero Presenter Stadlant.



1. 1 S.C. 1979



EBE # 27#3

Miguru 4. Akto briation Goufficiant var Hrvanwian Goodfon Misginedy, Anbanato Flow, karo teranna Gradian



AFWL-TR-71-57

Turbulent Boundary Layer Conference, and is regarded as a top quality flat plate experiment. The excellent agreement between the Aerotherm model and these data, both in profile and drag prediction, is apparent from these figures. Comparisons with subsonic, blown boundary layer data taken at MIT by Kendall (reference 66) are shown in Figures 5 to 8. These data were essentially the basis for the Aerotherm wall law model, although comparisons with other data have been shown to be equally good. As with the unblown case, profile comparisons are excellent. Drag is underpredicted by about 10-20% in comparison to the reported $C_{\underline{f}}/2$ values for this case; however the prediction is well within the error bands associated with these values.

b. The Cebeci - Smith Model (reference 68)

In the Cebeci-Smith model, the boundary layer is also characterized by inner and outer regions. In the inner region, a mixing length approach is used, based on the Prandtl mixing length law as modified by Van Driest (reference 75). Cebeci and coworkers have extended and modified the basic Van Driest law to include pressure gradient, blowing, and variable turbulent Prandtl number effects. The inner region mixing length expression is

$$\ell = k_{\rm m} y [1 - \exp(-y/A)]$$
(13)

where

$$A = \frac{26\nu}{\sqrt{\frac{\tau_w}{\rho_w}}} \left(\frac{\rho_w}{\rho}\right)^{1/2} \left(\frac{\mu}{\mu_w}\right) \left(\frac{1}{N}\right)$$
(14)

$$N = \left\{ \frac{\mu}{\mu_{e}} \left(\frac{\rho_{e}}{\rho_{w}} \right)^{2} \frac{p^{+}}{v_{w}} \sqrt{\frac{\tau_{w}}{\rho_{w}}} \left[1 - \exp\left(11.8 \frac{\mu_{w}}{\mu} \frac{v_{w}}{\sqrt{\frac{\tau_{w}}{\rho_{w}}}} \right) \right]$$
(15)

$$+ \exp\left(11.8 \frac{\mu_{W}}{\mu} \frac{v_{W}}{\sqrt{\frac{\tau_{W}}{\rho_{W}}}}\right)^{1/2}$$

$$P^{+} = \frac{\frac{\nu}{u_{e}^{2}} \frac{du_{e}}{dx}}{\left(\frac{C_{f}}{2}\right)^{3/2}}$$
(16)

AFWL-TR-71-57



Figure 5. Velocity Ratio Profiles Kendall, Subsonic Flow, F = 0.00325











Figure 8. Skin Friction Coefficient vs. Streamwise Location Kendall, Subsonic Flow, F = 0.00325



NENE-TH-71-57

k = Frendtl constant, taken as 0.40

The effect of blowing is reflected in the value of N, which in turn effects the damping constant A. Increasing the amount of blowing decreases A, which results in less damping, or a more rapid approach to the Prandel mixing length expressive.

In the outer region, a constant addy viscosity expression similar to equation (7) is used:

This addy viscosity is soltiplied by Elebanoff's (reference 76) interrittency factor.

$$T = \left[1 - 5.5 \left(\frac{T}{T}\right)^{6}\right]^{-1}$$
(15)

where f is the boundary layer thickness. As is the Accothern model, the inner region c_{μ} expression is used until it worsely the rater region expression, At which point the outer region c_{μ} is used.

The Cabaci model for addy conductivity is based upon an examination of the one-dimensional monsteady emergy equation subject to an oscillatory boundary condition, similar to his demivation of eddy viscosity. This analysis results in a concumptant model for variablest Pranéth number:

$$Pr_{z} = \frac{k_{z}[1 - exp(-y/h)]}{k_{y}[1 - exp(-y/h)]}$$
(25)

where

$$3 = \frac{34v}{\sqrt{\frac{1}{p_v}}} \left\{ -\frac{p^*}{v_v} \sqrt{\frac{1}{p_v}} \left[\exp\left(11.\frac{v_v}{\sqrt{\frac{1}{p_v}}}\right) - 1 \right] + \exp\left(11.\frac{v_v}{\sqrt{\frac{1}{p_v}}}\right)^{\frac{1}{2}} \right\} \right\}$$
(21)

 kg = an emergy mixing length constant, taken as 0.44
 (22)

 Pr = molecular Prandtl number
 (23)

APNL-TR-31-57

Since Sebeci does not consider combonogeneous flows, he does not have a model for turbulent Schwidt mumber.

The Cebeci-Smith turbulent model validity is well documented is numerous open literature publications (references 77 to 79).

c. Bushnell and Deckrith Model (reference 59).

Sushnell and Berkwith use a mixing length model for the entire boundary layer. Near the wall (to y/i = 4.1). Prandtl's mixing length expression modified by Van Driest's damping function is used:

:25

VOETE

local shear and desirity are included rather than wall values. The A* value is taken from a fit of experimental data on blown, subscule boundary layers, 15 shown in Figure 3. The experimental data are those of Simpson, et al. (reference 33) and the data reported in reference 2.

) = 100 5

In the far wall region (y/t > 0.3), mixing length is assumed to be a function it soundary layer thickness and the incompressible form tactor, $E_1^* = i\frac{\pi}{2}/F_1$, where

 $\dot{z}_{\pm} = \int_{0}^{2} \frac{z}{z_{e}} \left(1 - \frac{z}{z_{e}}\right) dy \qquad (25)$

Figure 10 shows the curve used to determine 4/3. This curve is the "quadratic variation" of reference 69, which has been adopted as the standard model at XASA/Langley (reference 80). The use of an 1/3 variation as a function of incompressible form factor is an attempt to account for nonequilibrium* effects in the profile shape. Thus, it is nost useful in strongly nonsimilar flow situations.

A straight line segment for 1/6 is used to join the inner and far wall regions between $y/\delta = 0.1$ and $y/\delta = 0.3$.

Nonequilibrium as used here refers to the fluid mechanical state rather than chemical state.



Figure 9. Variation of Damping Constant for Bushnell-Backwith Wall Mcdel



Figure 10. Variation of Mixing Length for Bushnell-Beckwith Wake Model

APML-TR-71-57

Bushnell and Beckwith use a constant turbulent Prandtl number but base it on total enthalpy fluctuations rather than static enthalpy fluctuations. That is,

$$\mathbf{Pr}_{T} = \frac{\varepsilon_{T}}{\varepsilon_{H_{T}}} = \left(\frac{\overline{(ov)}^{T} \overline{u}}{(ov)^{T} H_{T}^{T}}\right) \left(\frac{\partial \overline{H}_{T}}{\partial \overline{u}}\right)$$
(27)

Since no species equations are solved, no turbulent Schmidt number is specified.

The validity of the Bushnell-Beckwith model under several different kinds of flow condition: has been presented in reference 69.

2. TURBULENT MODEL COMPARISONS FOR THE LAW OF THE WALL REGION

a. WALAW Program

In order to examine the differences between the near wall mixing length formulations, a small computer code (WALAW) was written which solves the one-dimensional continuity, momentum, and energy equations. These equations, written for the wall region where streamwise changes are small compared to changes normal to the wall, are

continuity

 $\rho \mathbf{v} = \rho_{\mathbf{v}} \mathbf{v}_{\mathbf{w}} \tag{28}$

momentum

$$\rho_{\mathbf{w}} \mathbf{v}_{\mathbf{w}} \mathbf{u} = \rho \left(\mathbf{u} + \varepsilon_{\mathbf{m}} \right) \frac{\mathrm{d}\mathbf{u}}{\mathrm{d}\mathbf{y}} - \tau_{\mathbf{w}}$$
(29)

energy

$$\rho_{\mathbf{w}}\mathbf{v}_{\mathbf{w}}\left(\mathbf{h} + \frac{\mathbf{u}^{2}}{2} - \mathbf{h}_{\mathbf{w}}\right) = \rho\left(\mathbf{v} + \varepsilon_{m}\right) \frac{d\left(\mathbf{u}^{2}/2\right)}{d\mathbf{y}} + \left(\mathbf{k} + \frac{\rho\varepsilon_{m}c_{p}}{Pr_{t}}\right)\frac{d\mathbf{T}}{d\mathbf{y}} - \mathbf{q}_{\mathbf{w}} \quad (30)$$

A perfect, ideal gas equation of state is assumed, and Runge-Kutta integration is used. Inputs to the code are τ_w , q_w , T_w , and $(\rho v)_w$. Also, a law for the eddy viscosity variation must be included in the set of equations. The code includes the three wall region eddy viscosity relations described above, however a constant turbulent Prandtl number equal to 0.9 has been assumed in all three models.

b. Results for Three Near Wall Models

The differences between the three near-wall mixing length models are exemplified by the curves of Figures 11 through 15. The WALAW code was run for a 50 ft/sec, adiabatic flat plate flow with three different blowing rates, $\rho_w v_w / \rho_e u_e = 0.0, 0.004$, and 0.010. The turbulent models in these figures are labeled as follows:

- A = Aerotherm model
- B = Bushnell-Beckwith model
- C = Cebeci-Smith model

In Figure 11 where no blowing is included it is seen that all three models result in nearly the same mixing length variations. The Prandtl mixing length equation is essentially matched at about $y^+ = 100$. In Figure 12 where a moderate-tostrong blowing rate of 0.004 has been input, all three l^+ curves have shifted toward the Prandtl law, but the Bushnell and Cebeci models show more shift than the Aerotherm model. In Figure 13 where a very strong blowing rate of 0.010 is included, Models B and C have adopted the Prandtl line, $l = k_m y$, over the entire range of interest. Model A shows a similar trend with blowing, but does not respond as loadily. Thus, it can be expected from these results that, under no blowing conditions, there will be very little difference between near wall predictions with the three mixing length models. As blowing increases, differences will become apparent. When constrained to the <u>same wall shear value</u>, the Aerotherm law will give lower turbulence levels which in turn result in larger gradients ($\partial u/\partial y$, $\partial T/\partial y$ etc.) than the Cebeci and Bushnell models.

These conclusions about the similarities between predictions for unblown flows are borne out by the WALAW results of Figure 14 through 19. Figure 14 includes a velocity profile comparison with the Wieghardt and Tillman data (reference 38). All three models show good agreement in the "law of the wall" region, which extends to about 0.5 inch for the station selected. Beyond that point, the one-dimensional equations of motion are no longer valid. Figure 15 includes comparisons with Coles (reference 8) Mach 3.7 adiabatic flat plate data, which shows how well these laws apply to compressible flows. At least for adiabatic flows, any of the three models appears to be satisfactory in the supersonic range.

For hypersoni flows, the experimental data uncertainties become greater since the flow _ self is more difficult to work with. Figure 16 includes predictions from the three theoretical models compared to an experimental Mach number profile measured by Lee, et al. (reference 17). Mach number comparisons are shown since this information can be calculated directly from the pitot tube



÷,



Figure 11. Mixing Length Variation with No Blowing



Figure 12. Mixing Length Variation With F = 0.004



Figure 13. Mixing Length Variation With F = 0.010



Figure 14. Linear-Log Velocity Ratio Profiles Wieghardt, Subsonic Flow, Zero Pressure Gradient

AFWL-TR-71-57



Figure 15. 'inear Log Velccity Ratio Profile Coles Run #20, Supersonic, Zero Pressure Gradient Flow



MACH NUMBER, M

46

Linear-Log Mach Number Profile WOL, Lee, et al., Run 12902, Hypersonic, Zero Pressure Gradient Flow Figure 16.

AFWL-TR-71-57



Linear Log Static Temperature Profile NOL, Lee, et al., Run 12902, Hypersonic, Zero Pressure Gradient Flow Pigure 17.

47



Linear-Log Mach Number Frofile NCL, Lee et al., Run 6211, Hypersonic, Zero Pressure Gradient Flow

Figure 18.

MACA NUMBER, M



Linear-Log Static Temperature Profile NOL, Lee, et al., Run 6211, Hypersonic Zero Pressure Gradient Flow

Figure 19.

STATIC TEMPERATURE, OR

AFWL-TR-71-57

readings, thereby introducing the least possible instrument error. Skin friction values were measured using a floating element shear stress gage. As can be seen in figure 16, Mach number is underpredicted by about 15 percent by all the theories when this measured shear stress level is input to the WALAW program. A value of τ_w approximately 20 percent higher would give much better agreement, as shown in the figure. Similar results are obtained for the static temperature predicted wall shear; a τ_w value 20 percent larger gives more correct predictions with all theories considered. More will be said of the agreement with Lee's data later in this report.

Lee's higher stagnation pressure experiments show better agreement with theoretical predictions using the reported shear values. Figure 18 shows Mach number data and predictions for a $P_0 = 150$ psia run, the highest pressure reported by Lee. Agreement between Mach number data and profiles is excellent for all three theories, as are the static temperature predictions (Figure 19). The effect of a 20 percent change in wall heat flux is also shown in this figure, in order to establish the sensitivity of the one-dimensional solution to this input quantity. Thus, it can be concluded that for incompressible flows or compressible flows up to a Mach number of 5, with or without heat tr nufer, all three turbulent models do a satisfactory, and a very similar, job of predicting profiles in the near wall region.

As discussed earlier, Aerotherm's mixing length law has been verified by comparisons with Kendall's (reference 66) low-speed blown boundary layer data in another report (reference 2). These comparisons were rerun with the WALAW program to confirm their validity and also to obtain simultaneous predictions for the Cebeci and the Bushnell-Beckwith models. A small error was found in the original (reference 2) comparisons; however, the results remain essentially the same, as can be seen in Figures 20, 21, and 22. A11 results were obtained using the wall shear values reported by Kendall, determined by a wall pressure profile technique (as opposed to a $d\theta/dx$ technique). At the modest blowing rate of F = 0.002 (Figure 20), agreement between the Aerotherm and Cebeci models and the experimental data is excellent. The Bushnell-Beckwith model appears to react too strongly to the blowing rate, as svidenced by the very small extent of the laminar sublayer. A higher input $\tau_{\rm p}$ value would shift all predictions upward, however (see Figure 16). In Figure 21, results for a stronger blowing rate of F = 0.003 are shown. At this blowing rate, the Aerotherm model shows the least effect of blowing, while the Pushnell model shows the most. Agreement is good for Model A and Model C. Results for strong blowing are shown in Figure 23. Using







Linear-Log Velocity Ratio Profile MIT, Kendall Run C - 3 x 10⁻³ - 50, Low-Speed Flow With Blowing Figure 21.



Linear-Log Velocity Ratio Profile MIT, Kendall Run C - 5 x 10⁻³ - 50, Low Speed Flow With Blowing Figure 22.

AFWL-TE-71-57

AFWL-TR-71-57

the reported wall shear stress and blowing rate, orly the Aerotherm model predicts the velocity profile accurately. The Cebeci and Bushnell models result in nearly idencical profiles, giving an error of 50 percent in the law of the wall region. The uncertainty in wall shear can more than account for this entire error, however.

The Stanford experimental data (reference 33) provides an independent check on the low speed blowing results presented above, since the test conditions are nearly identical. Freestream velocities of approximately 50 ft/sec with no pressure gradient were used for the reference 33 experiments, with blowing rates of F = 0.0 to 0.008. Figure 23 illustrates the good agreement obtained between experiment and theory for a no-blowing case. Figures 24, 25, and 26 present data and predictions for nominal blowing rates of F = 0.002, 0.304, and 0.008, respectively. The theory trends are exactly as shown in the Kendall data comparisons. Models A and C show nearly the same results at low blowing, but diverge at stronger blowing rates. Models B and C give the same results with strong blowing. Using the reported wall shear values, the data in general follow the trends predicted by the Bushnell-Beckwith model more closely than the Cebeci model. Both of these models do a better job than the Aerotherm model. Thus, any conclusions which may have been drawn from the Kendall data comparisons must be questioned in the light of the Simpson comparisons. The difficulty lies in the requirement of an accurate C_f value, which is uncertain to + 100 percent or more for strongly blown flows.

One-dimensional theory and data comparisons were also carried out for the supersonic blown boundary layer experiments of Jeromin (reference 48). Jeromin experienced difficulties with axial pressure gradients and non-twodimensionality of the flow, therefore his data must be considered less dependable than the subsonic cases. Drag coefficients were inferred from the integral momentum equation, from Stevenson's wall law technique (reference 81), and by a transformation method. In the predictions for several of Jeromin's runs shown in Figures 27 to 30, a single τ_w value was used if the three C_f determination methods were in agreement. More than one τ_w value was used (and is illustrated in the figure) if the methods gave widely divergent results. Figure 27 shows the type of agreement obtained for an unblown, Mach 2.5 flow. Agreement with the predictions is fairly good at the lower indicated shear stress value, although not as good as the Coles' comparisons in supersonic flow (Figure 15). Moderate-to-strong blowing at this Mach number (Figure 28) gives poor agreement

^{*} The Kendall wall pressure profile technique for determining Cf was applied to several of Simpson's runs, but no identifiable trends in the alternate values for Cf were found.


Figure 23. Linear-Log Velocity Ratio Profile
Stanford, Simpson Run 2367-3, Low-Speed Flow Without Blowing



Figure 24. Linear-Log Velocity Ratio Profile Stanford, Simpson Run 122366-2, Low-Speed Flow With Blowing





AFWI.-TR-71-57









- | -^e

Figure 27. Linear-Log Velocity Ratio Profile Jeromin Run 2.5-0.0-1.40, Supersonic Flow Without Moving















AFWL-TR-71-57

with Model A and good agreement with Models B and C. Results for a Mach 3.5 flow are shown in Figures 29 and 30. With no blowing, agreement is again fairly good for all theories. A very strong blowing case was selected for the last comparison, Figure 30. As with the Simpson strong blowing case, agreement with Models B and C is good, while Model A gives poor agreement. However, an uncertainty in $\tau_{\rm u}$ of 100 percent for this case is not unlikely.

The preceding comparisons have shown that three leading turbulent models for near wall calculations are in substantial agreement for unblown and smallto-moderately blown flows. At larger blowing rates, with the wall shear level specified, the Aerotherm model is in substantial disagreement with the other two models and with the experimental data of Simpson a of Jeromin. On the other hand, the Aerotherm model is in agreement with the strong blowing experimental data of Kendall, which is not matched by the Cebeci and Bushnell theories. Thus, the choice of a "best" turbulent model is not at all clear.

The question of which model is best cannot actually be answered with these one-dimensional theory comparisons, since the predictions are so completely dependent on the wall shear rate. For blown boundary layers, the wall shear is generally highly uncertain. Thus, insufficient information is available to make a choice between the three models on purely technical comparison grounds. The basis for the choice which was made is presented in the next subsection.

3. THE SELECTED TURBULENT MODEL

a. Wall Law

The previous comparisons have centered on the wall law region, since this area of turbulence modeling is perhaps the most importan' and includes the widest disagreement between theories. The comparisons of Section III.2 have shown substantial agreement between theories for flows with no blowing or "small" blowing, but wide disagreement for flows with "strong" blowing.

In order to place the "small" blowing and "strong" blowing terminology in perspective, a practical example is presented here. Consider a 10° halfangle cone flying at sea level at Mach 10. Assume an ablating carbon or phenolic carbon surface on the cone, such that the wall temperature is about 6000° R. With all turbulent flow, the "no blowing" drag coefficients at the 1-foot and 5-foot body stations are approximately

> $C_f = 0.00042$ @ x = 1 foot $C_f = 0.00033$ @ x = 5 feet.

AFWL-TR-71-57

These values were calculated using the Schultz-Grunow formula corrected for compressibility:

$$C_{f_{i}} = \frac{0.37}{(\log_{10} \text{Re}_{x})^{2.50}}$$
(31)

$$\frac{c_{f}}{c_{f_{i}}} = \left(\frac{\rho^{\star}}{\rho_{e}}\right)^{0.8} \left(\frac{\mu^{\star}}{\mu_{e}}\right)^{0.2}$$
(32)

$$\frac{h^{*}}{h_{e}} = 1 + 0.5 \left(\frac{h_{w}}{h_{e}} - 1\right) + (0.22)r\left(\frac{h_{s}}{h_{e}} - 1\right)$$
(33)

For a carbon nose tip or heat shield, the ablation rate is characterized by plateau-like behavior at a $B'_C = \dot{m}_C / (\rho_e u_e C_H) = 0.2$ (reference 82). Thus, assuming $C_{\mu} = C_f / 2$,

$$\frac{(\rho v)_{W}}{(\rho u)_{e}} = 0.2 \left(\frac{C_{f}}{2}\right)$$
(34)

which results in a very small blowing rate. A similar behavior is obtained with phenolic carbon at a B_c of approximately 0.6. Thus, the blowing rates for typical heat shield materials are quite small under ordinary flight conditions.

In light of the fact that there are only small differences between models at zero or small blowing, and that small blowing conditions are typical for flight cases, the logical choice of a wall region turbulent model for BLIMP comparisons was the reference 1 model. The reference 1 model is already incorporated in the BLIMP code, which minimized the amount of programming time required. The constants within this model were held at the incompressible flow values of

$$k_{\rm m} = 0.44$$
 (35)

$$y_{2}^{\dagger} = 11.823$$
 (36)

The effects of changes in these constants will be demonstrated in Section IV.

b. <u>Wake Law</u>

There has been very little discussion in the literature about the advantages or disadvantages of a Clauser type wake law, as used in Models A and C compared to a boundary layer thickness dependent mixing length law, as used in Model B. All that seems to be required in the outer portions of equilibrium type flows is that a roughly correct eddy viscosity number be arrived at. This is borne out by the BLIMP results shown in rigure 31, where the constant in the expression

$$e_{m} = (const.) u_{e} \delta_{i}^{*}$$
(37)

was set at both 0.016 and 0.018, all other features of the reference 1 model remaining the same. This 12.5 percent change in the outer boundary layer turbulence coefficient over a distance of 2.75 feet in running length produced virtually no change in the velocity profile. The Clauser type formulation in terms of the kinematic or velocity displacement thickness δ_{1}^{*} , seems to be adequate for equilibrium compressible flows, therefore the reference 1 wake model was adopted for the final data comparisons using the 0.018 constant.

c. Turbulent Prandtl and Schmidt Number

The turbulent Prardtl and Schmidt numbers relate the turbulent transport of energy and chemical species to the turbulent transport of momentum. While there is only a weak basis for doing so, it is common practice to assume a constant ratio between the transport coefficients. Experimental data on actual turbulent Prandtl numbers and their variations through the boundary layer are becoming available (references 93, 84, and 85). The evidence indicates that \Pr_t should vary from ~ 0.5 in the outer portions of the boundary layer, to near unity at $y/\delta = 0.1$, to ~ 1.5 or 2.0 near the laminar sublayer. Results are primarily for air boundary layers. While these results are certainly relevant to the BLIMP turbulent model, the development and incorporation of a \Pr_t model which has reasonable validity for multicomponent chemically reacting flows was beyond the scope of the current effort. Thus, a constant \Pr_t model was used for the current studies. A similar situation exists for turbulent Schmidt number, although fewer data are available (e.g., reference 86).

Numerical studies with the BLIMP code and the reference 1 turbulent model have shown that a turbulent Prandtl number of 0.9 results in the generally accepted value of 0.88 for the flat plate recovery factor in air. These results, shown in Figure 32, were generated with BLIMP by demanding zero heat flux to the wall as a solution boundary condition. Thus, a constant value of $Pr_t = 0.9$ was used for boundary layer predictions with the current turbulent model.

AFWL-TR-71-57



Figure 31. Effect of Constant in Wake Region Eddy Viscosity Expression



Figure 32. Effect of Turbulent Prandtl Number on Recovery Factor at M = 3.7



SECTION IV

PREDICTIONS AND COMPARISONS WITH DATA

1. PRESENTATION OF COMPARISONS

The comparisons between BLIMP predictions and measured data for the 17 separate data sets listed in Table V are presented in this section. The results are listed in the order of Table V. They consist of (1) a brief description of the BLIMP input* as related to the data and (2) graphical results, where applicable, in the form of linear Mach number ratio, total temperature ratio, and velocity ratio profiles as functions of y, velocity ratio profiles as functions of y in linear-log coordinates, and linear plots of momentum thickness Reynolds number, Re_{A} , and skin friction coefficient, $C_{f}/2$, versus streamwise dimension, x. Symbolism has been standardized for all plots. Measured data are represented by circled points; where more than one set exists on a single plot, symbols are noted on that plot. BLIMP predictions started far upstream of the region of interest using an approximate starting profile (a "zero start") are presented as solid curves; predictions started at the first data station using an actual data profile (a "data start") are presented as dashed curves. Any deviations, modifications, or additions to this format are noted on the figure. In general, only one linear-log velocity plot is shown, usually the next to last of those plotted in linear coordinates.

a. Coles' Comparisons

(1) Comments on BLIMP Input

Coles' data consist of a single profile with C_f measurements at three other axial stations. Freestream and wall conditions were known to be effectively constant, thus BLIMP was started from the leading edge of the flat plate. The use of a fence trip, however, introduces an unknown initial thickness which is equivalent to the virtual origin being upstream of the leading edge. Since the data start option is not applicable with only a single station of profile data available, the only means of accounting for the trip effect was to match the profiles. The momentum thickness was selected as the appropriate

* A more complete discussion of computer code setup for these problems, including zero starts and data starts, is included in the appendix.

matching parameter. A BLIMP profile at the desired value of 0 was obtained by first running the program with sufficient stations to bracket the data value, then rerunning the program including the interpolated value of x as one of the output streamwise stations (see Figure 39). Thus the profile comparisons of Figures 33-38 compare data taken at x = 1.69 feet to predictions at 2.46 and 2.18 feet for M = 2.5 and 3.6, respectively.

(2) Comments on Results

Mach number and velocity ratio comparisons in Figures 33-38 for both runs are relatively good. Temperature profiles are not included since temperatures were not measured. In general, the prediction indicates higher values of Mach number and velocity in the mid-ranges of the profile and lower values beyond $u/u_{2} = 0.8$ to 0.9. There is a slight indication of an inflection point in the Mach number data profile which the prediction does not seem to include. Maximum absolute differences in ratio are 0.04 to 0.05. The plot of Re_A in Figure 39 is shown only to indicate the matching procedure. Finally the $C_{f}/2$ plot in Figure 40 shows the BLIMP predictions and data both referenced from the beginning of the plate. $C_{f}/2$ predictions for the θ -matched stations are 7.5 and 8.5 percent lower than the data for the Mach 2.5 and 3.6 cases, respectively. It should be noted, however, that improvements in the nodal distribution to be discussed in Section IV.2.b account for most or all of this error. Nodal spacing was discovered to be particularly important for θ -matched cases, in that a small improvement in $\frac{d\theta}{dx}$ brings about a new match point, which in turn gives a significantly different Cf.

b. NOL Comparisons

NOL data sets are comprised of profiles measured at four stations (Lee) and five stations (Brott) with skin friction measurements several inches upstream of each profile station. Measurements were taken over a streamwise interval from approximately 4 to 7.5 feet measured from the nozzle throat. Each profile was measured during a separate test run, thus stagnation conditions varied slightly (less than one percent) from profile to profile. Comparison to a single BLIMP run, which is based on constant stagnation conditions, is subject to some small error for this reason. In general, the BLIMP input of freestream conditions was based on the measured Mach number and average stagnation conditions for the four or five reported profiles. Wall temperatures were taken as the reported values. As a consequence of this approach, the pressure gradients at a given test station are slightly different for the prediction than actually existed. Axial pressure gradients are not discussed by



Figure 33. Mach Number Ratio Profile Coles Run #27, Supersonic, Zero Pressure Gradient Flow



Figure 34. Velocity Ratio Profile Coles Run #27, Supersonic, Zero Pressure Gradient Flow

AFWL-TP -71-57



Pigure 35. Linear-Log Velocity Ratio Profile
Coles Run #27, Supersonic, Zero Pressure Gradient Flow









a state and the second second of the second









•





Figure 40. Skin Friction Coeff cient vs Streamwise Location Coles Runs at Mach 2.6 and 3.7

and the second second second

Lee, et al., except to note that the 48-inch station was located in the acceleration region, 55 inches being the beginning of the constant freestrian flow where Mach number variations are constant within ± 0.75 percent. There is no indication of local gradient conditions at either the skin friction gages or profile station. Brott does list such data; however, there is currently no way to model the pressure gradients directly in the BLIMP code. Pressure gradients at the prediction stations are determined by a quadratic curve fit of the input pressure distribution which depends, of course, on the interval: between stations as well as the pressures.

- (1) Data of Lee, et al.
 - (a) Comments on BLIMP Input

To show the effects of pressure gradients, the 5 atmosphere run of Lee was set up as described above while constant freestream conditions equal to the average of those reported at the four stations were used for the 10 atmosphere run. Both cases were run with the data start option since conditions upstream of the first profile station were not reported. Several "approximate" zero start runs were attempted for the 10 atmosphere run, one using the nozzle profile reported by Brott and the other assuming constant conditions throughout. These results bracketed the measured values of θ only in a gross sense (see Figure 51). However, as shown in Figure 52, these approximate zero start runs do define limits on the value of $C_f/2$ within which a prediction based on the actual upstream conditions can be expected to fall.

(b) Comments on Results

Figures 41-46 and 47-51 show the results for the 5 and 10 atmosphere runs, respectively. In both cases, the predicted and measured profiles for Mach number, velocity and temperature diverge from the data as the flow proceeds down the plate (after the matched data start beginning). The predictions exceed the data in the mid-region of the Mach number and velocity profiles by up to 30 percent. In Figures 41 and 47 the Mach number data profiles appear to undergo a significant change in shape that is not reflected by the predictions.

In the Re_{θ} plots of Figures 45 and 51, the predicted growth, starting from the matched value at the first station, is less than that measured. This is consistent with the profile plots which show progressively "thicker"

^{*} Private communication with NOL personnel indicated these data were not measured.





TOTAL TEMPERATURE RATIO PROFILES NOL, ET AL., 5 ATM, HYPERSONIC, ZERO PRESSURE GRADIENT FLOM



.

۰.

•



Linear-Log Velocity Ratio Profiles NOL, Lee, et al., 5 Atm, Hypersonic, Zero Pressure Gradient Flow Figure 44.

こうち ちちちちちちちちち

ł

1

1









AFWL-TR-71-57

AFWL-TR .71-57





TOTAL TEMPERATURE RATIO PROFILES NOL, LEE, ET AL., 10 ATM, HYPERSONIC, ZERO PRESSURE GRADIENT FLON F16URE 48.

AFWI-TR-71-57





Linear-Log Velocity Ratio Profiles NOL, Lee, et al., 10 Atm., Hypersonic, Zero Pressure Gradient Flow Figure 50.

,





والمتحفظ فالمعاولة فالمحاور المقادلة المحاوي والمحاولة محاذة المتحد ومحاولا المحاولة المحاوية والمحاور والمحاور والمحاولة والمحاو



Skin Friction Coefficient vs Streamwise Location NOL, Lee, et al., 10 Atm, Hypersonic, Zero Pressure Gradient Flow Figure 52.
measured boundary layers in terms of θ even though the predicted and measured δ thicknesses remain essentially equal (see Figures 43 and 49). The consistency in the data appears doubtful when one considers the $C_f/2$ plots of Figures 46 and 52. For both runs, the $C_f/2$ predictions exceed the measured values after the first station (which is still within the accelerating region of the nozzle) by 16 to 24 percent. This is directly opposite to the trend seen in the Re_{θ} plots (Figures 45 and 51). Considering the momentum integral equation, which for zero pressure gradient and no blowing reduces to

$$\frac{C_{f}}{2} = \frac{d\theta}{dx}$$
(38)

it is clear that if $C_f/2$ is overpredicted, $\frac{d\theta}{dx}$ should also be overpredicted. The data, of course, result from skin friction gages, rather than a solution of the momentum integral equation. Calculation of $C_f/2$ using equation (38) and measured values of θ and x results in values between 8 and 9 x 10⁻¹ as compared to skin friction data between 3.6 and 4.0 x 10⁻⁴ in the uniform flow region. On the other hand, the BLIMP predicted $C_f/2$ and $d\theta/dx$ are equal.

In view of the above comparisons, the Lee data appear somewhat questionable. This is not to say the predictions are 100% correct, but only that internal inconsistencies in the data such as three-dimensional flow are suspected. Further comment will be made in Brott comparisons to follow.

The effect of the "input" pressure gradients in the Lee data is most apparent in the Re_{θ} and C_f/2 plots of Figure 45, 46, 51, and 52. In the first set (5 atm, Figures 45, 46) pressure variations are considered in the prediction while in the latter (Figures 51, 52) they are not. The axial pressure gradient distribution computed by BLIMP is negative between 3.77 and 4.7 feet, positive up to 7.0 feet and egative thereafter. The effect on Re_{θ} is seen in Figure 45; d θ /dx is la^{*} in regions of negative pressure gradient and smaller in the positive gradient regions. The momentum integral equation accounting for pressure gradients has the form

$$\frac{C_f}{2} = \frac{d\theta}{dx} - \frac{\theta}{\rho_e u_e^2} \frac{dp}{dx} \left\{ 2 + \frac{\delta^*}{\theta} - M_e^2 \right\}$$
(39)

* This behavior in Cf/2 is directly opposite that for subsonic flow due to the change of sign of the bracket term in equation 39 at high Mach numbers.

Table JIII lists the values of each of these terms at each of seven BLIMP output stations.

x (ft)	d 0 dx	Pressure Term	$\frac{C_{f}}{2}$
3.77	13.00 * 10 ⁴	8.25×10^{-4}	4.75 ± 10^{-4}
3.82	19.18	14.00	5.18
3.90	18.58	13.60	4.98
4.20	10.24	5.45	4.79
4.77	4.41	-0.32	4.73
5.77	0.04	-4.61	4.65
7.60	9.56	4.80	4.76

TABLE VIII

MAGNITUDES OF MOMENTUM INTEGRAL EQUATION TERMS

From Table VIII and Figure 46, it is apparent that the predicted value of $C_f/2$ experiences only minor fluctuations even though the value of the pressure term is oscillating with an amplitude several times the value of $C_f/2$. That is, the input pressure gradient is a very important term in the momentum integral equation; however, adjustments to account for its large variations occur in the θ growth rate $(d\theta/dx)$ rather than in $C_f/2$. The initial spike in $C_f/2$ is due to the incorrect input pressure gradient. The spike has no real significance since the second through the fourth stations were added only for purposes of the data start option, and the hand-interpolated pressure values resulted in locally high gradients. These pressure gradients offer no obstacle to the computer solution, and have no lasting effect on the downstream solutions; therefore, the solution was not rerun. The results do emphasize the experimental difficulties associated with accurate evaluation of local C_f values through use of the momentum integral equation

In Figure 51 the momentum growth is essentially linear as expected for a constant axial pressure and nearly constant C_f throughout the region. The variation in $C_f/2$ in Figure 52 is due to several factors. The initial rise is the rapid recovery of the profile next to the wall following the data start profile as described in the appendix. The slow variation up to x = 6.0 feet is believed to be due to the readjustment of the outer portion of the profile which does have a small effect on wall gradients. Downstream of x = 6.0 feet, this readjustment appears to be complete and $C_f/2$ decreases slowly as expected

for a uniform flow. These observations are further substantiated by comparison to the two solid curves which indicate the approximate zero start BLIMP predictions, assuming the upstream flow conditions noted.

- (2) Data of Brott, et al.
 - (a) Comments on BLIMP Input

The Mach number distribution from the throat to the first profile station is presented in reference 6 for the design and the measured values. Using this information, zero start BLIMP predictions were made assigning x = 0 at the nozzle throat. For additional comparison, "data start" runs starting from the first reported profiles at x = 47 inches were made. As with the 5 atmosphere Lee run above, freestream pressures were assigned to equate the local measured and input freestream to stagnation pressure ratios.

(b) Comments on Results

Figures 53-58 and 59-64 contain the results of the 5 and 10 atmosphere stagnation pressure runs, respectively. Results are plotted for the zero start and data start predictions described above. Profiles from the data start prediction for both runs tend to readjust in shape moving down the plate. By the final station, which is nearly 3 feet downstream from the first, these profiles are quite similar to the zero start profiles. This then is a measure of the duration of the recovery process. These figures show that the adjustment of the Mach number occurs more rapidly than that of total temperature and velocity.

In comparison with the data, both forms of the prediction tend to (1) indicate higher values of Mach number, total temperature, and velocity in the mid-range of the profiles up to Mach number ratios of 0.8 and velocity ratios of 0.9 and (2) underpredict in the far wake region, particularly the Mach number. As with Lee above, the boundary layer thicknesses, δ , are essentially equal.

The momentum thickness comparisons of Figures 57 and 63 are quite good for both starting conditions. Small differences in θ at the first data point are apparently due to differences between the Aerotherm and NOL curvefit of the profile data. This good overall agreement is in marked contrast to the results of the Lee comparisons. Figures 58 and 64 show excellent agreement between predictions and data for $C_f/2$. Note that there is no sudden change in $C_f/2$ after the first station for the data start predictions, indicating that the input, i.e., the measured wall g adient, was consistent with the BLIMP solution for the local conditions. The more gradual dips shown are probably due to



AFWL-TR-71-57











AFWL-TR-71-57

ī.

(//



Momentum Thickness Reynolds Number vs Streamwisc Lucation NOL, Brott, et al., 5 Atm, Hypersonic, Negative Pressure Gradient Flow Figur: 57.

AFWL-TR-71-57







MACH NUMBER CATTO PROFILES NOL, BROTT, ET AL., 10 ATM, HYPERSONIC, NEGATIVE PRESSURE GRADIENT FLOW





· , ·

•

•







Momentum Thickness Reynolds Number vs Streamwise Location NOL, Brott, et al., 10 Atm, Hypersonic, Negative Pressure Gradient Flow Figure 62.



ct/s × 10,

Skin Priction Coefficient vs Streamwise Location NOL, Brott, et al., 10 Atm, Hypersonic, Negative Pressure Gradient Flow Figure 64.

AFW1-TR-71-57

the overall profile adjustment taking place but may also be a result of the modeling of the pressure gradient in this region.

The log velocity plots of Figures 56 and 62 also show good agreement over the law of the wall region. The two data points nearest the wall in both of these figures are assumed to be in error since they are inconsistent with the skin friction balance data and the BLIMP prediction. These data points were taken with a very small probe (0.005 inch high) at very low pressures, and are subject to numerous sources of error. Referring back to the Lee log velocity plots (Figures 44 and 50), near wall measurements show better agreement with theory, perhaps since a larger (0.016 inch) probe was used. When the near wall readings from three stations are superimposed (Figures 44 and 50), the 5 atmosphere case shows good agreement while the 10 atmosphere case does not. This merely confirms the idea that the near wall data, even with the larger probe, are questionable. Farther away from the wall, over the bulk of the profile, the Lee data are in much poorer agreement with theory than Brott's data.

In conclusion, the Brott data appear to be much more consistent in terms of profile shape, θ growth, and reported $C_f/2$ values, both internally and in comparison to the predictions. The Lee data lack this internal consistency, particularly when $d\theta/dx$ is compared to the $C_f/2$ values. Also in the Lee experiments, the rapidly changing profile shape in an essentially "similar" flow region is disturbing.

c. <u>TRW Comparisons</u>

(1) Comments on BLIMP Input

Since tunnel operation was continuous, variations in stagnation and freestream conditions from profile to profile were quite small. Constant input conditions were determined as averages of the reported station values. Wall temperatures were input as measured.

An experimental difficulty that might have affected the profile data was the formation of a frost layer on the cold wall section. This was not detected until after the step-up run. A change was made in procedure, namely, data were taken during only the first 15 to 20 minutes once the wall was cooled. The model was then warmed and recooled until measurements were complete. This procedure was employed for the step-down tests. Due to the frost formation, Stanton tube data in cold wall regions were judged to be erroneous and thus, not reported.

^{*} Personal communication with Roland Lee at NOL indicated that correction curves for near wall measurements like these are currently being generated.

AFWL-TR-71-57

Figures 65 and 71 present the axial wall temperature distributions for the step-up and step-down runs. BLIMP comparisons were made at the stations marked by checks. For the zero start predictions, freestream and wall conditions upstream of the first reported station were assumed constant and equal to the values at the first station. In handling step changes in streamwise properties, BLIMP contains an option to treat streamwise derivatives as two point differences (linear variation) in place of the usual three point difference (quadratic curve fit). This option was applied to the step region from the first station in the step to the second station at the new wall temperature.

A data start run for the step-down case was made to model the profiles immediately upstream of the step exactly. In particular, accurate total temperature modeling was desired.

(2) Comments on Results

Profile and $C_f/2$ comparisons are presented in Figures 66-70 and 72-76 for the step-up and step-down cases. Step-down comparisons include both zero start and data start comparisons. No comparisons are shown for momentum thickness because the accuracy of these data are being reviewed by TRW. The five plotted comparison profiles include the last station before the step, three stations within the immediate region of the step, and the final station which was approximately 1 foot downstream from the step.

Profile comparisons for both cases are quite good, particularly for the Mach number and velocity. In the total temperature comparison for the step-up case (Figure 67), the prediction follows the recovery of the profile well but has a larger "bump" near the wall in the third and fourth stations shown. Some of this is due to the initial difference in the profiles at the first station, and some may be due to the resolution of the data, that is, only the open points represent actual temperature probe data. The closest approach to the wall was 0.050 inch; all solid points shown inside this were calculated from pressure probe data. A data start prediction would have modeled the data slightly better through the step; however, the differences were not regarded as serious enough to warrant further computer solutions.

The linear-log velocity ratio profiles in Figures 69 and 75 also reflect good agreement. Comparisons for both the first and final profiles are shown.

* Personal communication with Dr. Robert Gran at TRW.



from Leading Edge of Cylinder Wall Temperature vs X, Distance TRW Step-Up in Wall Temperature Figure 65.



FIGURE 66. MACH NUMBER RATIO PROFILES TRM, STEP-UP IN WALL TEMPERATURE, SUPERSONIC





Marken ...

ł

1.1.1.1.1

t

m

APH L-TR-71-57

P

Same Prove

La mere

ĺ.

The balance is a set of the set of the



Figure 69. Linear-Log Velocity Ratio Profiles TRW, Step-Up in Wall Temperature, Supersonic



ARWL-TR-71-57





AFWL-TR-71-57

- KALERANA



Wall Temperature vs X, Distance from Leading Edge of Cylinder TRW Step-Down in Wall Temperature Figure 71.





ちんん うだち







AFWL-TR-71-57





Figure 75. Linear-Log Velocity Ratio Profiles TRW, Step-Down in Wall Temperature, ^fupersonic





.....



119

AFWL-TR-71-57

2542-18-71-57

Note that staggered scales are used on the abcissa. Closest approach of the 5.618 inch thick total pressure probe was 6.009 inch, a distance well outside the predicted laminar sublayer.

Finally, the $C_f/2$ plots of Figures 70 and 76 show excellent agreement between the available warm-wall region Stanton tube and predicted values. The response of the $C_f/2$ prediction to the step follows the variation in wall temperature (Figures 65 and 71) with me noticeable lag, indicating the weak depen-'ence of the laminar sublayer on the owner portion of the profile.

d. Stanford Comparisons

(i) Comments on BLIMP Input

Freestream and wall conditions were maintained constant within several percent along the test section for all Simpson runs ased hereim. In three of these runs (no blowing, F = 0.002, and step-up in blowing), averaged constant streamnist pressures, wall temperatures and blowing mass flow rates were imput. For the other two runs, the actual reported values at each of the four stations were used. All predictions were started at X = 0 defined as the leading edge of the test section. The mirtual origin resulting from the use of a trip was located essentially at the leading edge of the test section.

(2) Comments on Results

Results of Stanford comparisons are plotted in Figures 77-90. Only velocity ratio profile comparisons are included since all tests chosen were essentially adiabatic at the wall with temperature variations arrows the boundary layer limited to a few percent. Comparisons are presented at each of the four stations reported. These are at staticns between x = 1.5 and x = 7.5 feet. Station locations are not the same for every test and are evted on each plot. For Figures 85 and 85, the step in blowing occurs at x = 5.02 fret. \Im_{4} and C_{f} comparisons are presented in a group (Figures 87-90) following the profiles. The four constant blowing runs are plotted together in Figures 27 and 90.

Overall profile snapes are in reasonably good agreement at all blowing rates (see Figures 77-85), however, the prediction indicates progressively higher values in the mid-range of the profile as the boundary layer proceeds downstream following quite good agreement at the initial station. (Due to this initial matching, data start runs were unnecessary.) This trend is confirmed in the Re₁ plots of Figures 87 and 83 which indicate slightly higher values of \pm for the measured data. It is also reflected in $C_{g}/2$ comparisons in Figures 39 and 90. As is well known, minor errors in \pm calculation can result in enormous errors in calculated drag coefficient for flows

WL-29-22-57

ş

ین از مین از مین از مینیند. رسین از مینین از مینین از مینین مینیند. از مینین از مینین از مینین مینیند. از مین مینین مینین مینین مینیند. مینین مینین مینین از مینین مینین مینین مینیند. از مینین مینین مینین مینین مینین مینین

: . ANTRIAL ANT

n na sensa na sensa sensa sensa kana na sensa sensa sensa sensa sensa sensa sensa sensa sensa sensa sensa sens A na sensa manda manya sensa sensa A manya ana sa tan sensa senta <mark>de bahaman</mark>ak dan sa sa sa sa sa sa sa sa sa sa

.





121



C. MARINE



Figure 78. Linear-Log Velocity Ratio Profiles Stanford, Simpson Run #2367, Subsonic, Nc Elowing













AFWL-TR-71-57*



Figure 82. Linear-Log Velocity Patio Profiles Stanford, Simpson Run #122066, Subsonic, F = 0.004
AFWL-TR-71-57



127

FIGURE 83. VELOCITY RATIO PROFILES STANFORD, SINPSON RUN #121966, SUBSONIC, F = 0.008





, a ,







Figure 86. Linear-Log Velocity Ratio Profiles Stanford, Simpson Run #5867, Step-Up in Blowing





Figure 87. Momentum Thickness Reynolds Number vs Streamwise Location Stanford, Uniform Blowing Runs, Subsonic

. 1

and the second division of the second divisio

and the second



Figure at. Momentum Stickholds Reynolds Number va Arreanwise bocation Btanfrid, Binyach Run Stato, Subsunte, Biep-Up in Blewing

2PG-11-71-57





F





AFWL-TR-71-57

with strong blowing. This holds true both for the reduction of experimental data and for prediction procedures such as BLIMP. This point will be discussed in more detail later.

The vertical bands on each data point represent the reported uncertainty intervals.* The differences between data and prediction in terms of percent of the data value are listed in Table IX.

*	
Simpson Run	Difference in $C_{f}^{/2}$
F = 0.000	12% low
F = 0.002	25% low
$\mathbf{F} = 0.004$	40% low
$\mathbf{F} = 0.008$	80% low
$F = 0.000 \div 0.004$	12% + 40% low

C_/2 COMPARISONS FOR STANFORD RUNS

TABLE IX

The consistency of this pattern is completed by the comparison of the near wall region in Figures 78, 80, 82, 84, and 86. In every case, the data indicate higher gradients at the wall by roughly the percentages of Table VIII.

The step in blowing causes an immediate 50 percent decrease in predicted $C_f/2$ and within the following 2 feet (15-20 boundary layer thicknesses) has decreased to within 10 percent of the value for constant F = 0.004 blowing from the leading edge.

e. Jeromin Comparisons

(1) Comments on BLIMP Input

All Jeromin predictions utilized the data start option since no data were presented defining the flow conditions from the nozzle throat to the first measured profile. Freestream conditions at each station were based on matching measured Mach numbers; wall conditions (temperature and mass flow) were input as reported. Initial station velocity profiles were taken from Squire (reference 98). Temperature (enthalpy) profiles were computed from equation (1) in Section II.3.c.

^{*}These uncertainty intervals have been described as too narrow by Squire, reference 87.

(2) Commetts on Results

Only velocity profile comparisons are presented because profile data were reported in that reduced form. For the runs in question, total temperature profiles were not measured but were calculated from equation (1). This procedure was justified by Jeromin based on the close agreement of preliminary measurements with equation (1). Consequently, since the measured Mach number profiles and the presented velocity profiles are directly related; no attempt was made to convert back to the Mach number form.

Figures 91-94 and 97-100 contain the velocity profiles for the runs 2.5-0.0, 2.5-1.2, 3.6-0.0, and 3.6-2.1 in that order. Comparisons are presented at four of the five profile stations reported by Squire. These latter profiles represent one-half of the number of measured profiles. Jeromin reports complete boundary condition and thickness integral data at all measured stations, which covered a streamwise range from about x = 1.1 to x = 1.5 feet. The Re₀ plots in Figures 95 and 101 include measured data at all stations. A single $C_f/2$ value obtained via the momentum integral method (by Jeromin) is shown for each test in Figures 96 and 102.

All the velocity profile comparisons indicate a recurring pattern; the BLIMP profiles progressively show higher velocity ratios in the mid-range of the profile. By the final profile, differences in velocity ratios range from 0.05 to 0.07 with the maximum occurring between $u/u_e = 0.6$ to 0.8. These differences are only slightly greater for the blown profiles than for the unblown.

The profiles reported by Squire are the actual Jeromin profiles for only the Mach 3.6 runs. Squire reran the Mach 2.5 experiments. Consequently, the Squire profiles and the Jeromin flow and wall conditions and computed profile parameters are related only in a nominal sense. This appears to be the reason for the small difference in initial data start matching of the Re_{θ} values for Mach 2.5 in the Figure 95 as compared to the more exact initial matching in Figure 101 for Mach 3.6. The trends are correct, however, as the Mach 2.5 profiles of Figures 91-94 clearly indicate that the measured d θ /dx should be greater than that predicted.

The nonlinear variation of Re_{θ} for Mach 2.5 with blowing in Figure 95 results from the recorded but unintentional axial pressure variation in the experiment. The irregularity at the downstream end of the test section in the Re_{θ} data in Figure 101 (Mach 3.6 with blowing) is due to apparent "blowing off"



FIGURE 91. VELOCITY RATIO PROFILES JERGIIII RUI: 2.5-0.0, CUPERSOULC, 10 DLOHING



Figure 92. Linear-Log Velocity Ratio Profiles Jeromin Run 2.5-0.0, Supersonic, No Blowing

AFWL-TR-71-57



139

FIGURE 93. VELOCITY RATIO PROFILES JEROMIN RUN 2.5-1.2, SUPERSONIC, F = 0.0012 AFWL-TR-71-57









AFWL-TR-71-57



Figure 96. Skin Friction Coefficient vs Streamwise Location Jeromin Mach 2.5 Runs





FIGURE 97, VILLOCITY RAJIO PROFILES JEROMIN RUH 3.6-0.0, SUPERSONIC, NO BLOWING -







:

;

:

÷

ALC: ABBUNC

VELOCITY RATIO PHOFILES JERONIN RUN 3.6-2.1. SUPERSONIC, F = 0.0021 FIGURE 99.

}6







Figure 101. Momentum Thickness Reynolds Nv wher ws Streamwise Location Jeromin Mach 3.6 Runs

AFWL-TR-71-57



Figure 102. Skin Friction Coefficients vs Streamwise Location Jeromin Mach 3.6 Runs

of the boundary layer as discussed by both Jeromin and Squire. Difficulties in obtaining pitot pressure data near the wall under these conditions are blamed for the unusual shape of the final data profile (x = 1.428 feet) in Figure 99.

The $C_f/2$ comparisons of Figures 96 and 102 substantiate the profile and Re₀ comparisons in indicating lower predicted values for blowing and no-blowing. The estimated uncertainty intervals reported by Jeromin are included. The familiar data start patterns are evident again. There is an immediate change to the approximate "equilibrium" value desired by BLIMP, followed by a gradual variation caused by a combination of axial pressure gradients and, probably the more significant factor, adjustments in outer profile shape. Comparative variations in the differences between measured and predicted $C_f/2$ and $d\theta/dx$ values relative to the measured values are given in Table X.

TABLE X

VARIATIONS IN MOMENTUM INTEGRAL TERMS FOR JEROMIN COMPARISONS

Run	C _f /2	d0/dx
2.5 - 0.0 2.5 - 1.2	18% low 45% low	15% lcw
3.5 - 0.0 3.5 - 2.1	15% low 90% low	28% low

Values of $d\theta/dx$ are included only for the no-blowing cases where, neglecting the effect of pressure gradients, equation (38) applies. As discussed previously, the presence of pressure gradients does not have a significant effect on the BLIMP prediction but as noted by Jeromin, it is significant in the determination of $C_{\rm f}/2$ from the momentum integral equation. This is particularly true with blowing when both $d\theta/dx$ and F are an order of magnitude greater than $C_{\rm f}/2$. Considering the pressure gradients and the three-dimensional effects reported by Jeromin, combined with the difficulties in accurately measuring $d\theta/dx$, it appears likely that the reported uncertainty intervals for blowing are underestimated.

One additional prediction of $C_{f}/2$ is shown for Mach 3.6 with blowing on Figure 102 as a solid curve. A zero start case was run from the nozzle throat

AFWL-TR-71-57

assuming streamwise conditions based on descriptions of the nozzle and the physical dimensions of the phrous plate in references 48 and 88. The matching of θ at the first station was about 20 percent off, but the $d\theta/dx$ from that point on was essentially identical to that of the data start prediction. The close agreement of the solid and dashed curves again confirms the ability of the data start procedure to respond to local wall conditions.

Figure 103 shows a comparison of static temperatures at the final station for M = 3.6 and no blowing. This is presented to assess the relationships among the temperature profile predicted by BLIMP (dashed curve), the temperature calculated from the Crocco relation, using the velocity profile predicted by BLIMP (triangles), and the measured Jeromin profile (circles). The close comparison between the two BLIMP determined temperatures indicates that the solution is in close agreement with the Crocco relation, and thus, in agreement with Jeromin's observations. The difference noted between Jeromin data and BLIMP predictions is consistent with the difference in the predicted and measured velocity profiles for x = 1.428 feet in Figure 97.

2. RATIONALIZATION OF RESULTS

Very lite effort was expended to improve the turbulent model for each data set as it was run, for two reasons. First, it would be unwise to make changes without evidence from a number of cases that a change was called for. Second, the contract for this study did not call for such an optimization. Rather, once a model was selected, it was to be evaluated for all data sets. Now that all the comparisons have been completed, however, it seems appropriate to examine the results, suggest what improvements to make, and how to make them.

a. Overview of the Agreement Between Experiment and Theory

All five of the data sets which have been used for comparisons here include data for unblown, essentially zero pressure gradient flows. In addition, the comparison with the data of Wieghardt and Tillman provided a baseline case with which the others can be compared. The velocity profile comparison was given in each case and in general was very good. Only the Lee case offered any significant errors in predicted velocity profile shape; however, the validity of the Lee data is open to question. Mach number profiles for the supersonic and hypersonic data sets are not predicted as well as velocity profiles. The problem is generally one of failing to predict an inflection point near the midrange of the Mach number ratio. In addition, M/M_e approaches 1.0 with a much greater slope than the prediction shows, which of course is related to the inflection point problem.

AFWL-TR-71-57



Figure 103. Static Temperature Profiles, Jeromin, M = 3.6, F = 0.000

Total temperature profiles are fairly good for NOL and TRW data, the only cases where such data were presented. The response of the prediction to the step in wall temperature in the TRW data was particularly encouraging. The $C_{f}/2$ and Re_{A} vs. streamwise distance plots are related through the momentum integral equation. There seems to be a general trend to underpredict $C_{f}/2$ and b variations slightly in several of the comparisons. This trend is disappointing in light of the excellent agreement with the Wieghardt and Tillman data. Some of this error can be eliminated by optimizing the nodal layout, as will be discussed in the next subsection, however it appears that the prediction will remain 2-3% low for the better quality data, such as Coles and Stanford. Referring back to the comparison between models of Figure 11, it is clear that all three turbulent models considered at the beginning of this report should give about the same results for unblown, low speed flows, with the Aerotherm model giving a slightly higher shear stress. Since the Cebeci model in particular has been shown to be very successful in predicting C_f for unblown flows (references 68, 77, and 78), it is hypothesized that the error observed here is a random one, and does not indicate a trend associated with the model. More will be said of this later.

For the blown flow data comparisons, velocity profile predictions are again quite good. In the Stanford data, there again seems to be a trend toward underpredicting the momentum thickness. This results in a gross underprediction of the reported drag coefficient for flows with moderate to strong blowing. This result could be anticipated from the one-dimensional analysis comparisons of Figures 24, 25, and 26. It was clear that, for the Aerotherm model, both the wall shear and the profile shape could not be predicted simultaneously. Thus, either the Simpson data are incorrect, or the Aerotherm model should be adjusted to fit it. This point is discussed further in the next subsection.

b. Changes in the Turbulent Model

For unblown flows, it has been stated above that the Aerotherm model is essentially equivalent to other, apparently successful models and there is no reason why it should not offer equivalent accuracy. A significant improvement in accuracy can be made by working with more nodes through the boundary layer, as discussed in the appendix. As more experience was gained in working with the code through the course of this contract, it became apparent that, in addition to those comments made in the appendix, the nodal distribution in the transition region (figure 194) is of primary importance. Figure 105



Figure 104. Diagram Showing Law of the Hall Nomenclature

shows $C_f/2$ for a 25-node model and for two 15-node runs, one with more emphasis placed on the transition region with correspondingly less on the wake region. This weighted model is in better agreement with the 25-node run which placed all the extra nodes into the transition and law of the wall regions. Pigure 105 also shows $C_f/2$ values calculated from several popular theories at $Re_x = 10^6$. It is apparent that there is a certain amount of disagreement over the correct drag coefficient value even for a simple, low speed, incompressible, flat flat problem.

It is of incerest to note that the weighted 15-node curve is nearly equal to the standard 15-node curve at the lower Reynolds numbers but approaches the 25-node curve at the higher Reynolds numbers. This is apparently due to the changing shape of the profile relative to the fixed $\overline{\eta}$ distribution. At the lower Reynolds numbers, the profiles are more nearly laminar with transition occurring in the outer portion of the $\overline{\eta}$ values. The 2⁵-node model has a sufficient number of nodes to model the transition region at any of the Reynolds numbers considered. It is obvious that the 15-node model lacks this flexibility, once more emphasizing the fact that to obtain the most accurate prediction, it is necessary to evaluate the results with respect to the chosen $\overline{\eta}$ distribution and to select that distribution based upon the streamwise region of greatest interest.

These comparisons were made after the results described in Section IV were obtained and plotted. Since the differences are small percentagewise, predictions were not rerun with the improved \overline{n} distributions. Thus, all zero blowing $C_f/2$ predictions would appear to be about 5% low for this reason.

One obvious way to alter the turbulent model is to change the numerical values of the constants. As an indication of how the constants would affect unblown boundary layer predictions, numerical experiments were performed for

AFWL-TR-71-57

No. of Concession, Name



Comparison of Predicted Skin Friction Coefficient for Zero Blowing Stanford Case to Data and Empirical Values Figure 105.



the Coles N = 3.7 case. Alternate values of both k_1 and y_2^+ were tried for this case, with the result shown in Figures 106 and 107. The small improvement in $C_{f}/2$ prediction for both the y_{a}^{+} and k_{m} modifications was made, at the particular station selected, with a corresponding small improvement in profile. These improvements are, of course, intimately related to the modal distribution through the transition region, therefore it is possible that the results may change at much larger axial distances. It was demonstrated in Section III.3.6 that small changes in the wake law edgy viscosity constant also result in small but detectable profile changes for unblown flows. Based on this evidence, it appears that hasic profile shapes and features are unlikely to be changed with modest changes in the model constants. "Pinc tuning" of the turbulent model to match drag or heat transfer data, for example, may be accomplished with small adjustments in these constants. Changes should be based on more comparison information than presented here, however. For flows at low Reynolds number or with high heat transfer rates, larger changes in these constants may be in order. The results presented here will be useful in estimating the results of such changes.

For flows with strong blowing, the effects of constant changes are altered somewhat. The wall law constants, y_a^+ and k_m , have the greatest influence in the transition region of the profile, which is much nearer the wall with strong blowing. Thus, there is virtually no change in the outer profile shape for different wall law constants, as shown in Figure 108. Significant differences near the wall do affect the drag coefficient, however, as seen in Figure 109. Manipulation of the y_a^+ type constant is essentially the technique used by both Cebeci and Bushnell and Beckwith to account for blowing in their models, therefore this path does appear to be a strong possibility if blown flow model changes are indeed desired. The question of whether such model changes are in order is addressed in the next subsection.

c. Desirability of Turbulent Model Changes for Blown Flows

A considerable body of data and numerous theories now exist for turbulent boundary layers with injection. The discussion, evaluation, and reevaluation, of this data has been a favorite topic in the recent fluid mechanics literature. Of particular interest for purposes of the present discussion is the drag coefficient correction due to blowing at any given point on a flat plate. Figure 110, taken from reforence 89, presents a number of theoretical solutions to this problem for incompressible flows, along with a few points from Simpson's data. The current Aerotherm theory essentially



c¹/s × 104

Figure 106. Effect of Wall Law Changes on Drag Coefficient in Unblown Flat Plate Flow

٠,











F. ure 108. Effect of Wall Law Changes on Velocity Profile in Flat Plate Flow With Blowing



Figure 109. Effect of Wall Law Changes on Drag Coefficient in Flat Plate Flow With Blowing



.





AFWL-TR-71-57

والمعامل والمرابع والم

duplicates the reference 2 curve. It is clear that Simpson's data shows less C_f correction than any of the theories presented. Figure 111, taken from reference 87, illustrates the point that Simpson's data show less C_f correction than that found by other experimentalists. Thus, while the Stanford heat and mass transfer apparatus has been very carefully constructed and operated by competent researchers, there is not universal agreement that the drag data obtained with this apparatus are correct. This merely is a result of the fact that with present measuring techniques, the calculation of drag by either momentum integral or wall profile techniques in blown flows is not sufficiently precise to draw any accurate quantitative conclusions.

It has been shown in Section III.3 that strong blowing is not encountered in typical heatshield or nosetip flight calculations. It is therefore concluded that changes in the Aerotherm wall law model are not warranted until more conclusive experimental data in the strong blowing region are available.



c+\s × 104



Comparison of Blowing Correction From Several Investigators, Taken From Reference 87 Figure 111.

.
SFCTION V

CONCLUSIONS AND RECOMMENDATIONS

Results of this research program are summarized briefly and conclusions are drawn in Section V.1. Recommendations for further work are included in Section V.2.

1. CONCLUSIONS

The research and development program described in this report has uncovered useful information about the state of the art in turbulent boundary layer experimentation and turbulence modeling. It has also answered many questions about the use of BLIMP as a prediction tool. In the litewature survey portion of this contract, it became evident that no one set of experimental data is completely suitable for turbulent model studies, i.e., completely error free. There is significant disagreement between various data sets for even very straightforward experiments, such as flat plate subsonic flow. Experimental problems and potential errors are compounded for the more difficult cases, such as flows with blowing.

Comparisons of three leading methods of modeling turbulence in boundary layers showed that the methods were very similar in unblown flows, but contained potentially important differences for flows with blowing. The one-dimensional solution technique which was used to compare these models appears to be useful for further development of turbulent models and possibly for screening new experimental data.

The comparisons of predictions and data presented in this report are valuable in establishing the degree of confidence which should be placed in BLIMP predictions. In general, it can be concluded that the eddy viscositytype model for turbulence is satisfactory for the type of flows considered here. BLIMP does a good job of predicting velocity and temperature profiles for a wide variety of flows and boundary conditions. Some improvements in profile shape are possible, however, particularly for higher Mach number flows. Of particular interest in the data profile shape is an inflection point in velocity and Mach number profiles which occurs at supersonic and hypersonic speeds. Some changes in the turbulent model, possibly including a variable turbulent Prandtl number, may be necessary to model this particular profile feature.

Drag coefficient predictions for many of the no-blowing cases considered in this report were slightly low. This was found to be a function of the number of nodes used and/or their spacing through the boundary layer. It can be concluded that the use of 15 nodes is very near the lower limit for accurate turbulent boundary layer predictions, and that strong consideration should be given to up-dimensioning the code to 25 nodes.

For boundary layer flows with blowing, profiles were again good. Drag coefficients were typically below the reported values for both the Simpson and the Jeromin data. With the Simpson data, there is considerable disagreement in the literature as to whether the reported drag coefficients are correct. For this reason, the rather poor agreement with the BLIMP predictions could only be termed "consistent but inconclusive." Since strong blowing is not typically encountered in heatshield ablation problems, it is concluded that a change in the turbulent model to fit the Simpson or Jeromin drag data is not justified without further study.

The very large number of computer runs necessary for the preparation of accurate predictions has resulted in new information on the use of the BLIMP code. A technique for starting a problem with a known profile at the first station was developed, and the "data start" runs emphasized some interesting features of the downstream solution. The general conclusion to be made from the data start runs is that wall region profiles (and the associated wall shear) approach the zero start predictions very rapidly, with the outer profile taking somewhat longer. This result then provides some information as to the accuracy inherent in the usual technique of starting a solution far upstream of the region of interest, with the expectation that starting profile errors will die out quickly.

The comparisons included in this report include demonstrations of the sensitivity of the predicted profiles to different numerical values of the turbulent model constants. It is concluded that, for unblown boundary layers, small (<30%) changes in these constants will not have any important effects on profile shapes. On the other hand, for flows with blowing, changes in the constants can have large effects on the profiles very near the wall, and therefore affect drag, heat transfer, etc. Thus, changes in the model constants as a function of blowing rate offers a straightforward method of altering wall parameters with this turbulent model. It may also be concluded that the model would be sensitive to other changes such as the substitution of τ_w for τ in equation 6.

2. RECOMMENDATIONS

Many areas for further investigation have become apparent during this study. Perhaps the foremost of these is the need for a straightforward continuation of the kind of work reported here, i.e., documentation of the validity of the code through comparison with experimental data. Fine tuning of the turbulent model for unblown flows should be carried out in order to make BLIMP the accurate and sophisticated prediction tool which it is intended to be. Initial studies should concentrate on drag data, then be extended to heat transfer. Much of the data screening and model development work could be carried out most efficiently with a one-dimensional code such as the WALAW program described in this report.

Once this fine tuning phase is completed, attention should be given to other types of flows not covered in the present study. Among the many flow regimes which merit attention are flows with large heat transfer, low Reynolds numbers (near transition), chemically reacting flows, and flows in adverse pressure gradient. All these conditions exist at the surface of a reentry vehicle, where the code is used to predict the resulting boundary layer. The need for verification is obvious.

The question of drag prediction in flows with blowing should also be resolved, perhaps through comparison with wall heat transfer rather than drag data. As a minimum, comparisons with some of the other cases shown in Table II should be carried out.

Finally, the need for additional experimental data in all types of flows is apparent. The hypersonic boundary layer area is of most interest for reentry vehicle purposes. Research programs aimed at the development of new instrumentation for use in blown or ablating boundary layer flows are particularly needed in order to eliminate the uncertainties that were brought out in this report.

165/166

APPENDIX

GENERAL DISCUSSION OF COMPUTER CODE SETUP AND OPTIONS

This appendix contains information on the use of the BLIMP code for the type of problems encountered with the selected data sets. It also presents details on the new entropy layer option.

1. SELECTION OF NODAL (n) DISTRIBUTION

The BLIMP solution procedure operates in the $(\xi,\overline{\eta})$ coordinate system, where these quantities are defined as

$$\mathbf{g} = \int_{0}^{\mathbf{g}} \rho_1 \mathbf{u}_1 \mu_1 \mathbf{r}_0^{2\kappa} d\mathbf{s}$$
 (40)

$$\overline{\eta} = \frac{u_1}{\alpha_{II}} \int_{0}^{Y} \rho r^{\kappa} dy \qquad (41)$$

A $(\xi,\overline{\eta})$ grid system is assumed to be superimposed on the boundary layer region, where $\overline{\eta}$ is measured normal to the wall and ξ is measured parallel to it (see Figure 112). The boundary layer is divided into n-1 strips connecting N nodal points at each ξ station. These nodal points are designated by $\overline{\eta}_i$ where i = 1 at the wall and N at the edge of the velocity boundary layer. The nodal system expands and contracts with the boundary layer flow, since $\overline{\eta}_1$ is defined to be located at the wall, and $\overline{\eta}_N$ is defined to be the outer edge of the boundary layer. While the ξ numerical values are calculated automatically by the program from the axial station dimension (s), the $\overline{\eta}_1$ numerical values which it uses are supplied directly as input. This subsection provides some insight into the selection of $\overline{\eta}_i$ values for proper program operation.

a. Number of Nodes Required

Since BLINF solves a linear matrix of order proportional to N (the number of \overline{n} nodes), the time required to obtain a solution can be expected to be roughly proportional to N squared. Consequently, it is desirable

^{*} An average of the matrix inversion $(-N^3)$ and other operations which are proportional to N and N².



Figure 112. \overline{n} - ξ Coordinate System

to determine the minimum number of nodes (and spacing of those nodes) which is consistent with an accurate and stable calculation of the boundary layer. No attempt was made to evaluate the minimum number aspect of this question; however on a number of different runs of Coles, Stanford and others, two BLIMP runs were made which were identical with the exception of the number of nodes, 15 being used on one and 25 on the other.

The results of one of the runs which are representative of all those made are as follows. First, for runs of Coles' test #20 with the same 8 axial stations, the number of iterations to a solution at each station was the same for either 15 or 25 nodes. The time required per iteration averaged 0.47 seconds for the 15-node run compared to 1.46 seconds for the 25-node run, a ratio of 0.31. The ratio of squares is 0.36; that is, a 25-node iteration took slightly longer than estimated by the N-squared proportionality. The differences in skin friction coefficient, C_f , and momentum thickness, θ , were consistently about 2.5 percent; the 15-node run having the lower values. Other comparison runs indicated similar differences (0 to 5 percent in C_f and θ) with the 15 node model giving consistently lower values.

Figure 113 shows the two velocity ratio profiles on a linear-log scale with the 25-node run as a solid line and the 15 node as circles. This evidence together with that above is judged sufficient to conclude that the 15-node model represents a worthwhile saving in computer time while maintaining satisfactory accuracy. Consequently, this model was used extensively for making the final BLIMP predictions. Due to the smaller number of nodes available, however, careful judgement had to be exercised in the choice of the nodal distribution.

b. Distribution of Nodes

Figure 114 depicts a typical turbulent velocity profile and also the variation of the first derivative of velocity through the boundary layer. The velocity gradient typically decreases three to four orders of magnitude between the wall value and the value at $u/u_e = 0.9$. Thus, the <u>a priori</u> selection of a nodal spacing to "curvefit" these variations with ten or fifteen discrete values is a difficult problem.







Figure 114. Typical Velocity and Velocity Gradient Profiles in a Turbulent Boundary Layer

The BLIMP manual (reference 90) suggests that the nodal spacing should be such that each successive value of \bar{n} not exceed the previous value by much more than a factor of 2. This guideline has proven to be generally valid but should be evaluated relative to each particular type of profile. As profiles become distorted, as in the case of blowing, it may be necessary to warp the distribution as well to be certain that regions of high gradients are vepresented adequately. For curvefitting purposes, it is undesirable to have a change of greater than 0.1 in the velocity ratio between any adjacent nodes. If experimental data are available, a quick study of reported velocity profiles will enable the user to select a satisfactory distribution. In the event data are unavailable, a short computer run limited to a few stations can confirm the adequatey of the selected distribution or indicate necessary changes.

Perhaps the most important part of the distribution is that nearest the wall. Since BLIMP calculates C_f from the wall velocity gradient, and since the gradient at the wall is taken as the first derivative of the first spline fit quadratic evaluated at y = 0 (see reference 1), it is essential that the first several nodal points be located within the laminar sublayer. As a rule of thumb, at least the first two points away from the wall should have velocity ratios less than 0.1. This together with the general spacing guideline above should result in satisfactory prediction of the various wall parameters.

Another less obvious region of concern is the outer edge of the boundary layer. The properties of the quadratic and cubic spline-fits (the outer two points are joint i by a cubic) are such that injudicious nodal spacing near the outer edge can cause an oscillatory overshoot of the edge value of 1.000 as shown in Figure 115. In this example the second to last node has been



Figure 115. Overshoot of the Spline Fit Procedure

chosen as the fixed node (generally defined as $u/u_e = 0.95$ and $\overline{\eta} = 1.000$). The next to last node may have been chosen too <u>close</u> to the fixed node and/or the last node too <u>far</u> from the fixed node. Since the cubic is constrained to $u/u_e = 1.00$ and $\partial (u/u_e)/\partial \overline{\eta} = 0.00$ at the final point (in the absence of an entropy gradient), it is possible to generate an overshoot. Such an occurrence is especially troublesome since integral properties are computed by an exact integration of the quadratic and cubic curve segments. If the differences in $\overline{\eta}$ are large, even small excursions in u/u_e may lead to large errors in the various thickness integrals. Avoidance of this problem is again possible by observing simple guidelines.

First, if a data profile is available, the spacing of the final nodes can be approximated well enough. For subsonic flew, \overline{n} and y are directly proportional. Supersonic flow requires somewhat smaller \overline{n} spacing due to the (generally) decreasing density. Flows with blowing tend to approach the edge condition with higher gradients (du/dy) and thus require smaller spacing compared to unblown flows. If a profile is not available, a trial run may be necessary using the generalized appro. Generalized below. Special attention should be given to the values of FP (F FRIME = u/u_e) and to FPP (F DOUBLE PRIME = $\partial(u/u_e)/\partial\overline{n}$) at the last few points, particularly those points between the fixed point and the final point. If any of the points has a velocity ratio very near or exceeding 1.00 and/or if there is an inflection or change of sign in the derivative, an overshoot may have occurred. It is always helpful to plot these two values over the final (ew points on a linear scale to be sure.

During the course of this investigation the following choices for the $\overline{\eta}$ distribution were made and were found to provide satisfactory results. The selection of which node should be the fixed \overline{n} node, the \overline{n} value for that node and the velocity ratio at that node is arbitrary. Choosing the value of \overline{n} as 1.00 makes for ease in ratioing to other values and is traditionally accepted, as is the choice of 0.95 for the velocity ratio (turbulent flow only). Both have been used exclusively for final BLIMP runs presented in this report. The number of the fixed node has been chosen as the 13th out of 15 total nodes. In some earlier runs, the 12th was fixed; however, this increases the possibility and the amplitude of overshoot due to a poor choice of sparing and additionally, places more nodes than necessary in a region of the profile that does not require as much detail. The next to last node should be chosen so as to result in a velocity ratio of 0.380 + 0.005. This will minimize the possibility of overshoot. Once again, an available profile is the best guide. The values in Table XI which ere used for this contract may be used as a starting point if profiles are not available.

TABLE XI

TYPICAL	ñ	DISTRIBUTIONS.	IN	THE	OUTER	WAKE	REGION	

	14th n P	oint	15th n Point		
Type of Flow	Range of Values	Mean Value	Range of Values	Mean Value	
Subsonic, no blowing	1.3	1.3	1.8	1.8	
Subsonic, blowing	1.2 + 1.3	1.25	1.5 + 1.7	1.6	
Supersonic, no blowing	1.3 + 1.5	1.4	1.8 + 2.7	2.1	
Supersonic, blowing	1.15 + 1.2	1.2	1.67 + 1.75	1.7	
Hypersonic	1.5 + 1.7	1.6	2.5 + 3.0	2.7	

One final point should be considered. As the solution procedure progresses axially in the flow direction, the \overline{n} grid system is "stretched" in order to remain fixed to the outer edge of the growing boundary layer. Since the laminar sublayer does not grow as rapidly as the turbulent outer flow, the grid points nearest the wall may eventually be stretched out of the laminar region altogether, thereby giving inaccurate wall gradient information. It is necessary therefore to check the results at all important stations to be certain that the \overline{n} distribution is sufficient, especially at the wall. Restarting the program at some intermediate body station is a possibility for very long running lengths.

2. THE "DATA START" PROCEDURE

In the classical boundary layer problem, the initial and boundary conditions are such that a known profile (or profiles) is provided at an upstream station, and sufficient edge and wall conditions are provided along the flow direction. Profiles are then found at body stations of interest based on the boundary conditions and upstream profiles. In typical AFWL applications of the BLIMP code, however, upstream profile information is not available, and the program has been written to start the solution based on a similar solution profile at the first station (usually close to X = 0). When started in this manner sufficiently far upstream of the region of interest, the errors resulting from the approximate profile die out and accurate solutions are obtained at downstream locations. For most of the data sets chosen, edge and wall informaticn were available in sufficient detail and accuracy to enable starting BLIMP in the conventional manner, that is, with a similar solution profile at an axial station slightly greater than zero (chosen here as 0.01 feet). In the supersonic and hypersonic tunnel flows where the test surface forms one side of the nozzle, conditions through the acceleration section are often not defined. Such was the case with Jeromin, Lee, et al., and to some degree with Brott, et al. An alternate procedure to enable comparisons to be made is to start BLIMP in the more classical manner, i.e., at the first reported data profile station with the measured velocity and temperature (enthalpy) profiles. The correct implementation of this starting procedure requires some detailed consideration, however

The BLIMP code instructions as described in reference 90 do indicate that the program can be started with an "input" profile as the solution at the firsu station. This starting option was intended primarily for restarting purposes, wherein an actual BLIMP solution is used as input to the code for further calculations. As an actual solution, unis input profile satisfies the spline fit

quadratic and cubic relations between nodes which are inherent in the integral matrix formulation. Thus, to use this starting option successfully with actual experimental profiles at the first station, the spline fit requirement must also be satisfied. For the "data start" cases presented in this report, another computer program* was used which accepted the measured velocity and temperature profiles in physical coordinates, converted them to the \overline{n} coordinate system, performed a least squares spline-type curve fit for 15 preselected \overline{n} nodal values, and printed out the standard BLIMP "restart" information. This restart information was then provided to BLIMP as the first station profile. All data start runs reported herein include this technique and in all cases convergence was achieved at the second station in a normal manner, that is, in from 3 to 8 iterations with an average of only four iterations.

A few comments regarding the preparation of data and the resulting performance of the data start procedure are in order at this point to establish the degree of accuracy which has been attained in the use of this technique. Since none of the profiles of present interest are sufficiently defined near the wall, it is necessary to generate additional points for input to the least squares curve fit program. Figure 116 shows a typical case: the data of Jeromin run 2.5-1.2 (Mach 2.5, F = 0.0012). Only the portion of the profile near the wall is shown; the circles represent the reported data and the dashed line represents the value of $d(u/u_p)/dy$ at the wall based on the reported value of C_f . The solid curve represents an "eyeball" curve fit from which additional points were chosen. The necessity of this is apparent considering that at least two values of \overline{r} not including $\overline{n} = 0.0$ should be chosen with $u/u_{a} < 0.1$ and about 3 more chosen in the remaining interval up to the first reported data point away from the wall. Any other regions which may not contain a sufficient number of data points may be "filled in" in a similar manner. The resulting supplemented experimental data constitute the required input profile. The temperature profile is obtained in the same manner except in Jerowin's cases, wherein temperature was defined by the velocity profile.

It is also of interest to examine the solution at nearby stations once BLIMP accepts the data start profile and proceeds on downstream. Figure 117 shows the progression in profile shape for the Jeromin case mentioned above. The solid curve with circles indicates the input velocity profile. The subsequent change in the solution for each nodal point is shown by the various symbols noted in the legend (including the station location). These solutions

^{*} This program was on hand at Aerotherm and was not developed under this contract.



Figure 116. Definition of Near Wall Velocity Profile for Data Start Input

176



³n/n

AFWL-TR-71-57

are also connected by lines indicating the loci of the nodal point solutions. The nodal points of the final calculated profile are connected by the dashed curve. From comparisons made on other cases, this dashed curve would be quite similar to the profile shape BLIMP would predict based on a zero start.

At the second axial station (triangle symbols), there is little if any change over the bulk of the profile; however, near the wall major adjustments have occurred. This is due to a difference between the input $C_f (d(u/u_p)/dy)$ at the wall) and the value BLIMP would predict given the same free stream and wall conditions. This adjustment takes place over very small axial distances. It appears to be identical to adjustments which are made in response to steps in wall conditions such as temperature and blowing rate. Although the magnitude of the adjustment is large, it is limited to a very thin layer near the wall. It therefore has a negligible effect upon thickness integrals which change in a smooth and continuous manner (see Section IV). As the solution continues on downstream, an adjustment is made to the outer portion of the law of the wall region while the overall "normal" growth of the boundary layer is reflected by a steady thickcning out in the wake region. By the final dashed profile shown, the adjustments to the shape appear to be complete, and ordinary boundary layer growth accounts for any shifting of the points. A number of examples of the manifestations of this behavior are apparent in the graphical presentations in Section IV. In general, "recovery" from the data start profile is (1) essentially immediate for wall proporties such as C_f , (2) for all practical purposes, unaffected by the starting values of the various integral parameters and (3) slow for the overall profile, taking on the order of 10 boundary layer thicknesses. This latter "recovering" simply implies that, given the existing conditions at the first reported profile station, BLIMP would not hat predicted that profile, the difference in shape being similar to that between the solid and dashed profiles in Figure 117.

3. HOMOGENEOUS FLOW CONSIDERATIONS

All of the cases considered in this turbulent model study involved only air as a working fluid, both as the main strcam gas and as the transpirant. Also, the temperature range under consideration was low enough that no chemical reactions would take place. Thus, it is possible to realize some economies in the operation of the BLIMP program in this limited thermochemical regime. The program was modified to operate in a homogeneous flow mode by accepting a single species in the thermochemical data deck. This species is treated as an element (named "cold air"); it is given its own fictatious atomic number and the usual set of thermochemical curvefit constants. Accurate transport

properties can also be realized in this mode of operation by reading in correct diffusion factor data under Group 12 of the program input.

4. ENTROPY LAYER OPTION

A new feature of the BLIMP code is the entropy layer option, which offers a direct coupling of entropy gradients in the inviscid flow with the boundary layer edge conditions. Inviscid flow entropy can be determined directly as a function of ϕ , the dimensional stream function. Further this functional relation can, for the case of a typical reentry vehicle, be determined from the shock shape. The entropy is established from the shock angle, and the stream function from the simple integration of a uniform freestream flow..

If the boundary layer "swallows" a sufficient flow mass to result in sizeable entropy variations within the swallowed mass, it is necessary to account for this phenomena adequately. An iterative mass balancing procedure is used to establish the correct edge entropy. Given an initial estimate of the edge condition, a solution is generated. The edge stream function from this solution demands a certain edge entropy, which is then compared with the edge entropy calculated from the edge pressure and enthalpy. These two entropies may be different, in which case the estimate of edge conditions is altered and a new solution is obtained. This iteration procedure is an integral part of the existing solution iteration procedure, and therefore does not generally add to the number of iterations required or the program run time.

The fact that the BLIMP code is written in terms of normalized dependent variables adds some confusion to the entropy layer solution. It is well known that the existence of an entropy gradient at the edge of the boundary layer also requires a velocity gradient, i.e.,

$$\frac{\partial u}{\partial y} = \frac{\partial a}{\partial y}$$
(42)

Since velocity varies in the inviscid flow at the edge of the boundary layer, the definition of a u value to be used in forming the dimensionless velocity ratio u/u is difficult. The problem was solved by using a reference velocity, u_r , defined by an <u>isent copic</u> expansion from stagnation conditions. The formulation of the equations as carried out in reference 1 remains valid with the new stipulation that

$$\frac{f'}{\alpha_{\rm H}} = \frac{u}{u_{\rm r}} \tag{43}$$

The pressure gradient parameter β enters the nondimensional formulation of the momentum equation in the same way, since the Bernoulli equation holds for the isentropic expansion

$$-\frac{1}{\rho}\frac{\partial P}{\partial x} = u_{r}\frac{\partial u_{r}}{\partial x}$$
(44)

and we will define

$$\beta \equiv 2 \frac{\partial \ln u_r}{\partial \ln \xi}$$
(45)

The pressure gradient term in the momentum equation remains essentially unaffected as

$$\beta\left(\frac{\rho_{\rm r}}{\rho}-f^{\prime 2}\right)$$

In the BLIMP solution procedure, the value of $u_{e}/u_{r} = f_{N}^{*}/\alpha_{H}$ is a variable at each station for entropy layer flows. It is determined from straightforward energy relationships for the reference and actual inviscid expansions around the body. As with the isentropic edge condition, the selection of the solution domain (i.e., the selection of the maximum value of \overline{n}) is arbitrary. Conventional techniques for interpreting the results in this solution domain must be re-examined, however. For example, at any given body station, two individual computer runs with different choices for the numerical value of \overline{n}_{N} would result in two different edge velocity values. Different values of u_{e}/u_{r} would also occur. Both solutions are correct, however, since a velocity gradient should exist in an entropy layer region. This velocity gradient will project one edge state to the other and the choice of the \overline{n}_{N} value merely determines how far into the inviscid region the boundary layer solution will extind.

One other interesting feature of the entropy layer operation of the program is the definition of the coordinate stretching parameter, $a_{\rm H}$. Formerly, coordinate stretching was accomplished by constraining some arbitrary point near the boundary layer edge, $\bar{n}_{\rm c}$, to have a specified velocity ratio, C, near

(but something less than) the edge value (reference 1). With an edge velocity gradient, this constraint has been modified to deal with the straight line extrapolation of the edge velocity, rather than the ratio itself, as shown in Figure 118. The selection of kappa and CBAP input values (Group 4, Card 3 reference 90) is not changed.





REFERENCES

- Anderson, L. W. and Xendall, R. M.; <u>Λ Nonsimilar Solution for Multicomponent</u> <u>Reacting Laminar and Turbulent Boundary Layer Flows Including Transverse</u> <u>Curvature</u>, Air Force Weapons Laboratory, Kirtland Air Force Base, New <u>Mexico</u>, Report AFWL-TR-69-106, March 1970.
- Kendall, R. M., Rubesin, M. W., Dahm, T. J., and Mendenhall, M. R.; <u>Mass</u>, <u>Momentum</u>, and Heat Transfer Within a Turbulent Boundary Layer with Foreign Gas Mass Transfer at the Surface, Part I - Constant Fluid Properties, Vidya Division, Itek Corporation, Final Report No. 111, 1964.
- 3. Adcock, J. B., Peterson, J. B., and McRee, D. I.; Experimental Investigations of a Turbulent Boundary Layer at Mach 6, High Reynolds Numbers, and Zero Heat Transfer, NASA TN D-2907, July 1965.
- 4. Allen, J. M.; Experimental Preston Tube and Law of the Wall Study of Turbulent Skin Friction on Axisymmetric Bodies at Supersonic Speeds, NASA TN D-5660, February 1970.
- 5. Brinich, P. F. and Neumann, H. E.; "Some Effects of Acceleration on the Turbulent Boundary Layer," <u>AIAA</u> Journal, Volume 8, No. 5, May 1970, pp. 987-989.
- Brott, D. L., Yanta, W. J., Voisinet, R. L. and Lee, R. E.; <u>An Experimental Investigation of the Compressible Turbulent Boundary Layer with a Favorable Pressure Gradient</u>, Naval Ordnance Laboratory, White Oak, Maryland, NOLTR 69-143, August 25, 1969.
- Cary, A. M.; "Turbulent Boundary Layer Heat Transfer and Transition Measurements for Cold Wall Conditions at Mach 6," <u>AIAA Journal</u>, <u>Volume</u> 6, No. 5, May 1968, pp. 958-959.
- 8. Coles, D. E.; <u>Measurements in the Boundary Layer on a Smooth Flat Plate in</u> Supersonic Flow. III. <u>Measurements in a Flat Plate Boundary Layer at the Jet</u> <u>Propulsion Laboratory</u>, Jet Propulsion Laboratory, California Institute of Technology, Pasadena, California, Report No. 20-71, June 1953.
- 9. Escudier, M. P.; The Turbulent Incompressible Hydrodynamic Boundary Layer, Ph.D. Dissertation, Imperial College of Science and Technology, University of London, London, England, October 1957.
- Fischer, M. C., Maddalon, D. V., Weinstein, L. M., and Wagner, R. D.; Boundary Layer Surveys on a Nozzle Wall at N=20 Including Hot Wire Fluctuation Measurements, AIAA Paper No. 70-746, June 1970.
- Graber, B. C., Weber, H. E., and Softley, E. J.; <u>Comparison of Lamina: and</u> <u>Turbulent Cone Boundary Layer Flow With and Without Pressure Gradient</u>, General Electric Company, Missile and Space Division, Philadelphia, Pennsylvania, Report R67SD49, November 1967.
- Hill, F. K.; "Turbulent Boundary Layer Measurements at Mach Numbers from 8 to 10," <u>The Physics of Fluids</u>, Vol. 2, No. 6, November-December, 1959, pp. 668-690.

- Hopkins, R. A. and Nerem, R. M.: "An Experimental Investigation of Heat Transfer from a Highly Cooled Turbulent Boundary Layer," <u>AIAA Journal, Vol.</u> 6, No. 10, Oct. 1968, pp. 1912-1918.
- 1⁴. Jones, R. A. and Feller, W. V.; <u>Preliminary Surveys of the Wall Boundary</u> Layer in a Mach 6 Axisymmetric Tunnel, NASA TN D-5620, February, 1970.
- 15. Julien. H. L., Kays, W. M., and Moffat. R. J.; The Turbulent Boundary Layer on a Porous Plate, Experimental Study of the Effects of a Favorable Pressure Gradient, Department of Mechanical Engineering, Stanford University, Report HMT-4, April 1969.
- 16. Kearnly, D. W., Kays, W. M., Moffat, R. J., and Lloyd, R. J.; <u>The Effect of Free-Stream Turbulence on Neat Transfer to a Strongly Accelerated Turbulent Boundary Layer</u>, Department of Mechanical Engineering, Stanford University, Report. HMT-9, February 1970.
- 17. Lee, R. E., Yanta, W. A., and Leonas, A. C.; <u>Velocity Profile, Skin Friction</u> <u>Balance</u>, and <u>Heat Transfer Measurements of the Turbulent Boundary Layer at</u> <u>Mach 5 and Zero Pressure Gradient</u>, Naval Ordnance Laboratory, White Oak, <u>Maryland</u>, NOLTR 69-106, June 1969.
- 18. Unpublished test data received from John E. Lewis, TRW Systems, Fedondo Beach, Calif. Data groups 61-77, 134-157, and 200-244 of Project No. VA0084, SAMSO-TRW Turbulent Boundary Layer Study completed August 26, 1970.
- 19. Lloyd, R. J., Moffat, R. J., and Kays, W. M.; <u>The Turbulent Boundary Layer</u> on a Porous Plate, An Experimental Study of the Fluid Dynamics with Strong Favorable Pressure Gradients and Blowing, Department of Mechanical Engineering, Stanford University, Report HMT-13, May 1970.
- 20. Lobb, R. K., Winkler, E. M., and Persh, J.: NOL Hypersonic Tunnel No. 4 Results, Part VII, Experimental Investigation of Turbulent Boundary Layers in Hypersonic Flow, U.S. Naval Ordnance Laboratory, White Oak, Maryland, NAVORD Report 3880, March 1955.
- 21. Matthews, R. K. and Trimmer, L. L.; Nozzle Turbulent Boundary Layer Measurements in the VKF 50 Inch Hypersonic Tunnels, Arnold Engineering and Development Corp., Tullahoma, Tenn., AEDC-TR-69-118, June 1969.
- 22. Metting, F. W., Chapman, D. R., Nykolm, J. R., and Thomas, A. G.; <u>Turbulent</u> <u>Skin Friction at High Mach Numbers and Reynolds Numbers in Air and Helium</u>, NASA TR-R-82, 1961.
- 23. Meier, H. U. and Rotta, J. C.; Experimental and Theoretical Investigations of Temperature Distributions in Supersonic Boundary Layers, AIAA Paper No. 70-744, June 1970.
- 24. Michel, R., Quemard, C. and Elena, M.; "Velocity Distributions in a Turbulent Boundary Layer in Uniform or Accelerated Compressible Flow," translation from la Recherche Aerospatiale, No. 128, January-February, 1969, pp. 33-47, NASA TT-F-12,538, August 1969.
- Moffat, R. J. and Kays, W. M.; <u>The Turbulent Boundary Layer on a Porou</u> <u>Place, Experimental Heat Transfer with Uniform Blowing and Suction</u>, Department of Mechanical Engineering, Stanford University, Report HMT-1, August 1967.
- Moore, D. R. and Harkness, J.; "Experimental Investigation of the Compressible Turbulent Boundary Layer at Very High Reyn Ids Numbers," <u>AIAA Journal</u>, Vol. 3, No. 4, April 1965, pp. 631-638.

- 27. Nagamatsu, H. T., Wisler, J. C., and Sheer, R. E.; Hypersonic Laminar and Turbulent Skin Friction and Heat Transfer on a Slender Cone, Ballistic Systems Division, Air Force Systems Command, Norton Air Force Base, California, BSD-TR 67-85, March 1967.
- 28. Neal, L.; A Study of the Pressure, Heat Transfer, and Skin Friction on Sharp and Blunt Flat Plates at Mach 6 8, NASA TN D-3312, April 1966.
- 29. Noonan, B. J. and Rand, J. L.; <u>High Temperature Turbulent Heat Transfer</u> <u>Measurements</u>, U. S. Naval Ordnance Laboratory, White Oak, Maryland, NOLTR 63-276, January 1964.
- 30. Perry, J. H. and East, R. A.; "Experimental Measurements of Cold Wall Turbulent Hypersonic Boundary Layers, In Hypersonic Boundary Layers and <u>Flow Fields</u>," Agard Conference Proceedings, No. 30, May 1968.
- 31. Samuels, R. D., Peterson, J. B., and Adcock, J. B.; Experimental Investigation of the Turbulent Boundary Layer at a Mach Number of 6 with Heat Transfer at High Reynolds Numbers, NASA TN D-3358, March 1967.
- 32. Scaggs, Norman E.; <u>Boundary Layer Profile Measurements in Hypersonic</u> <u>Nozzles</u>, Aerospace Research Laboratories, Wright Patterson AFB, Ohio, <u>APL-66-0141</u>, July 1966.
- 33. Simpson, R. L., Kays, W. M., and Moffat, R. J.; <u>The Turbulent Boundary Layer on a Porous Plate-An Experimental Study of the Fluid Dynamics with Injection and Suction</u>, Stanford University, Department of Mechanical Engineering, Report No. HMT-2, December 1967.
- 34. Thielkahr, ". H., Kays, W. M., and Moffat, R. J.; <u>The Turbulent Boundary</u> <u>Layer, Experimental Heat Transfer with Blowing, Suction, and Favorable</u> <u>Pressure Gradient</u>, Department of Mechanical Engineering, Stanford <u>University, Report HMT-5</u>, April 1969.
- 35. Wallace, J. E.; "Hypersonic Turbulent Boundary Layer Studies at Cold Wall Conditions," Proceedings of the 1967 Heat Transfer and Fluid Mechanics Institute, Stanford University Press, Stanford, California June 1967.
- 36. Wallace, J. E., <u>Hypersonic Turbulent Boundary Layer Measurements Using an</u> <u>Electron Beam</u>, Cornell Aeronautical Laboratory, Inc., Buffalo, New York, <u>Meport No. AN-2112-Y-1</u>, August 1968.
- 37. Whitten, D. G., Kays, W. M., and Moffat, R. J.; <u>The Turbuler's Boundary Layer on a Porous Plate, Experimental Heat Transfer with Variable Suction, Blowing, and Surface Temperature, Department of Mechanical Engineering, Stanford University, Report HMT-3, December 1957.</u>
- 38. Wieghardt, K. and Tillman, W.: U and M 5617 (1944), translated as On The Turbulent Friction Layer for Rising Pressure, NACA TM 1314, 1951.
- 39. Wilson, D. M.; Measurements of Hypersonic Turbulent Heat Transfer on a Highly Cooled Cone, U. S. Naval Ordnance Laboratory, White Oak, Maryland, NOLTR 67-24, July 1967.
- 40. Winkler, E. M. and Cha, M. H.; <u>Investigation of Flat Plate Hypersonic</u> <u>Turbulent Boundary Layers with Heat Transfer at a Mach Number of 5.2</u>, U. S. Naval Orgnance Laboratory, White Oak, Maryland, NAVORD Report 6631, Sept. 1959.
- Alsaji, S. J., <u>The Transpired Turbulent Boundary Layer with Heat Transfer</u>, Ph.D. Dissertation, <u>Mechanical Engineering Department</u>, University of Utah, Salt Lake City, Utah, June 1968.

- 42. Baker, E.: Influence of Mass Transfer on Surface Friction at & Porous Surface, Ph.D. Dissertation, Imperial College of Science and Technology, University of London, fondor, England, January 1967.
- 43. Bartle, E. R. and Leadon, B.M.: "The Effectiveness as a Universal Measure of Mass Transfer Cooling for a Turbulent Boundary Layer," Proceedings of the 1962 Heat Transfer and Fluid Mechanics Institute, Stanford University Press, Stanford, Calif., July 1967, pp. 2,-41.
- 44. Danberg, J.E.: <u>Measurement of the Characteristics of the Compressible</u> <u>Turbulent Boundary Layer with Air Injection</u>, U. S. Naval Ordnance Laboratory, White Oak, Maryland, NAVORD Report 6683, Sept. 1959.
- 45. Danberg, J. E.: <u>Characteristics of the Turbulent Boundary Layer with Heat</u> and <u>Mass Transfer at M = 6.7</u>, Naval Ordnance Laboratory, White Oak, Maryland, NOLTR 64-99, October 1964.
- 46. Dershin, H., Leonard, C. A. and Gallaher, W. H.: "Direct Measurement of Skin Friction on a Porous Flat Plate with Mass Injection," <u>AIAA Journal</u>, Vol. 5, No. 11, Nov. 1967, pp. 1934-1939.
- 47. Fernandez, F. L. and Zukoski, E. E.: "Experiments in Supersonic Turbulent Flow with Large Distributed Surface Injection," <u>AIAA Journal</u>, <u>Vol. 7</u>, No. 9, Sept. 1969, pp. 1759-1767.
- 48. Jeromin, L. O. F.: <u>An Experimental Investigation of the Compressible Tur-</u> bulent Boundary Layer with Air Injection, ARC R/M 3526, Nov. 1966.
- 49. Jones, J. W.; <u>A Turbulent Boundary Layer with Mass Addition, Combustion,</u> and Pressure Gradients, Ph.D. Dissertation, University of Utah, Salt Lake City, Utah, June 1970
- 50. Kent, R. Martellucci, A. and George, F.; <u>Static Force and Induced Pressure</u> <u>Measurements on a Slonder Porous Cone with Gas Injection</u>. (U), <u>Aeromechanics and Macarials Laboratery Operation</u>, <u>Re-Entry Systems Dept.</u>, <u>General Electric Company</u>, <u>Philadelphia</u>, <u>Pennsylvania</u>, <u>Aerodynamics Lab</u>. <u>Data Memo</u>, ALDM 69-75, February 1969.
- 51. Kuigein, N.G. "Transport Processes in a Combustible Turbulent Boundary Layer," Journal of Fluid Mechanics, 12, Part 3, March 1962, pp. 417-437.
- 52. Martellucci, A.; <u>Static Force</u>, <u>Surface Pressure</u>, and <u>Profile Measurements</u> on a Porous Cone with <u>Gas Injection - Data Presentation - General</u> Electric, Reentry and Environmental Systems Division, Valley Forge, Pa., Document No. 70SD462, <u>GAMSO TR-70-152</u>, <u>January</u> 1970.
- 53. McQuaid, J.; Experiments on Incompressible Turbulent Boundary Layers with Distributed Injection, ARC R/M-3549, Jan. 1967.
- 54. Muzzy, R. J.; "Surface Mass Addition into a Turbulent Boundary Layer," AIAA Journal, Vol. 5, No. 5, Nay 1967, pp. 1029-1033.
- Pappas, C. C. and Okuno, A. F.; "Measurements of Sk n Friction of the Compressible Turbulent Boundary Layer on a Cone with Foreign Gas Injection.," Journal of the Aerospace Sciences, Vol. 27, No. 5, May 1960 pp. 321-333.
- 56. Pappas, C. C. and Okuno, A. F., <u>Measurement of Heat Transfer and Recovery Factor of a Compressible Turbulent Boundary Layer on a Sharp Cone with with Foreign Gas Injection</u>, NASA TO 0-2210, April 1964.

- 57. Peterson, J. B., McRee, D. I., Adcock, J. B., and Braslow, A. L.; Further Investigation of Air Injection Through Slots and Porous Surfaces On Flat Plate Turbulent Skin Frictic. at Mach 3, NASA TN D-3311, March, 1966.
- 58. Romanenko, P. N. and Kharchenko, V. N.; "The Effect of Transverse Mass Flow on heat Transfer and Friction Drag in a Turbulent Flow of Compressible Gas Along an Arbitrarily Shaped Surface," Int. J. of Heat and Mass <u>Transfer</u>, <u>Vol.</u> 6, No. 8, August 1963, pp. 727-738.
- 59. Smith, T. H.; Hot Film Characteristics in a Turbulent Boundary Layer with Foreign Gas Injection, Ph.D. Dissertation, Mechanical Engineering Department, University of Utah, Salt Lake City, Utah, June 1969.
- 60. Squire, L. C.; Further Experimental Investigations of Compressible Turbulent Boundary Layers with Air Injection, ARC R/M 3627, August 1968.
- 6'. Wimberly, C. R., McGinnis, F. K., and Bertin, J. J.; "Transpiration and Film Cooling Effects for a Slender Cone in Hypersonic Flow.," <u>AIAA Journal</u>, <u>Vol. 8</u>, No. 6, June 1970.
- 62. Winkler, E. M., Madden, M. T., Humphrey, R. L., and Koenig, J. A ; Supersonic Ablation Studies with Terlon, Naval Ordnance Laboratory, NOLTR 69-125, Oct. 1969.
- 63 Wooldridge, C. E., and Muzzy, R. J.; "Boundary Layer Turbulence Measurements with Mass Addition and Combustion," <u>AIAA Journal</u>, Vol. 4, No. 11, Nov. 1966, pp. 2009-2016.
- 64. Kline, S. J. Morkovin, M. V., Sovran, G., and Cockrell, D. J., editors; Pro:eccings-Computation of furbulent Boundary Layers, 1968 AFOSR-IFP-Stanford Conference, Stanford University, Department of Mechanical Engineering, 1969.
- 65. Mickley, H. S. and Davis, R. S.; <u>Momentum Transfer for Flow Over a Flat</u> <u>Plate with Blowing</u>, NACA TN 4017, November, 1957.
- 66. Kendall, R. M.; Interaction of Mass and Momentum Transfer in the Turpulent Boundary Layer, Ph.D. Dissertation, Massachusetts Institute of Technology, Cambridge, Massachusetts, 1959.
- 67. Hahn, J. S. and Lutz, R. G.; Experimental Investigation of Turbulent Boundary Layers With Pressure Gradient and Heat Transfer at Mach 4, Arnold Engineering and Development Center, Tullahoma, Tennessee, AEDC-TP-71-3, January, 1971.
- 68. Cebeci, T.; Calculation of Compressible Turbulent Boundary Layers with Heat and Mass Transfer, AIAA Paper No. 70-741, June, 1970.
- 69. Bushnell, D. M. and Beckwith, I.E.: <u>Calculation of Nonequilibrium</u> Hypersonic Turbulent Boundary Layers and Comparisons with Experimental Data, AIAA Paper No. 69-684, June 1969.
- 70. Patankar, S. V. and Spalding, D. B., Heat and Mass Transfer in Boundary Layers, Morgan-Grampian, London, 1967.
- 71. Martellucci, A., Rie, H., and Sontowski, J. F.: Evaluation of Several Eddy Viscosity Models Through Comparison with Measurements in Hypersonic Flows, AIAA Paper No. 69-688, June, 1969.
- 72. Powell, T. E. and Strong, A. B. ; "Calculation of the Two-Dimensional Turbulent Boundary Layer with Nass Addition and Heat Transfer," in

Proceedings of the 1970 Heat Transfer and Fluid Mechanics Institute, T. Sarpkaya, editor, Stanford University Press, Stanford, California, 1970.

- 73. Reyhner, T.; "Finite Difference Solution of the Compressible Turbulent Boundary Layer Equations," in <u>Proceedings</u>, <u>Computation of Turbulent</u> <u>Boundary Layers - 1968 AFOSR-IFP-Stanford Conference</u>, S. J. Kline, et.al., editors, Stanford University, Department of Mechanical Engineering, 1969.
- 74. Herring, H. J. and Mellor, G. L.; <u>A Computer Program to Calculate Incom-</u> pressible Laminar and Turbulent Boundary Layer Development, NASA CR-1564, March 1970.
- 75. Van Driest, E. R.; "On Turbulent Flow Near a Wall., Journal of the Aerospace Sciences, Vol. 23, No. 11, November 1956.
- 76. Klebanoff, P. S.; Characteristics of Turbulence in a Boundary Layer with Zero Pressure Gradient, NACA TN 3178, July 1954.
- 77. Cebeci, T., Smith, A. M. O., and Mosinskis, G.; "Calculation of Compressible Adiabatic Turbulent Boundary Layers," <u>AIAA Journal</u>, <u>Vol. 8</u>, No. 11, November 1970.
- 78. Cebeci, T.; <u>A Model for Eddy Conductivity and Turbulent Prandtl Number</u>," McDonnell-Douglas Corporation, Long Beach, California, Report No. MDC-J0747/01, May 1970.
- 79. Cebeci, T. and Mosinskis, G. J.; Prediction of Turbulent Boundary Layers with Mass Addition, Including Highly Accelerating Flows, ASME Paper No. 70-HT/SPN-19, 1970.
- 80. Personal Communication with Dr. Dennis Bushnell, NASA Langley Research Canter, November 1970.
- 81. Stevenson, T. N.; <u>A Law of the Wall for Turbulent Soundary Layers with</u> Suction or Injection, The College of Aeronautics, Cranfield, England, Report Aero No. 166, July 1963.
- 82. Bartlett, Eugene P., and Grose, Ronald D.; <u>The Multicomponent Laminar</u> <u>Boundary Layer Over Graphite Sphere Cones, Solutions for Quasistendy</u> <u>Ablation and Application</u> to <u>Transient Reentry Trajectories</u>, Aerotherm Corporation, Mt. View, <u>Calif.</u>, Final Report No. 68-35, 29 May 1968.
- Simpson, R. L., Whitten, D. G., and Moffat, R. J., "An Experimental Study of the Turbulent Prandtl Number of Air with Injection and Suction," <u>Inter-</u> national Journal of Heat and Mass Transfer, <u>Volume 13</u>, 1970, pp. 125-143.
- 84. Meier, H. U. and Rotta, J. C.; Experimental und Theoretical Investigations of Temperature Distributions in Supersonic Boundary Layers, AIAA Paper No. 70-744, June 1970.
- Johnson, D. S., "Velocity and Temperature Electuation Measurements in a furbulent Boundary Layer Downstream of a Stephise Discontinuity in Wall Temperature," <u>Journal of Applied Mechanics</u>, <u>Jolume 26</u>, 1959, pp. 326-336.
- 86 Goldman, J. B. and Marchello, J. M.; "Turbulent Schmidt Numbers," <u>International Journal of Heat an. Mass Transfer</u>, Volume 12, 1969, pp. 797-802.
- 87. Squire, L. C.; "The Constant Property Turbulent Boundary Layer with Injection; A Reanalysis of Some Experimental Results," <u>International Journal of Peat and Mass Transfer</u>, Volume 13, No. 5, May 1970, pp. 939-947.

- 83. Squire, L. C.; Further Experimental Investigations of Compressible Turbulent Boundary Layers with Air Injection, Aeronautical Research Council, R and M No. 3627, August 1968.
- 89. Simpson, R. L., Moffat, R. J., and Kays, W. M.; "The Turbulent Boundary Layer on a Porous Plate: Experimental Skin Friction with Variable Injection and Suction," <u>International Journal of Heat and Mass Transfor</u>, <u>Volume 12</u>, No. 7, July 1969, pp. 771-789.
- 90. Anderson, L. W., Bartlett, E. P., and Kendall, R. M.; <u>Users' Manual</u>, <u>Boundary Layer Integral Matrix Procedure</u>, Air Force Weapons Laboratory, Kirtland Air Force Base, New Mexico, AFWL-TR-69-114, Volume I, March 1970.
- 91. Rubesin, M. W.; An Analytical Estimation of the Effect of Transpiration Cooling on the Heat Transfer and Skin Priction Characteristics of a Compressible Turbulent Boundary Layer, NACA Report No. TN 334:, 1954.
- 92. Mickley, H., Ross, R. C., Squyers, A. L , and Scewart, W. L.; Heat, Mass, and Momentum Transfer for Flow Over a Flat Flate With Blowing or Suction, NACA TN 3208, 1954.
- 93. Dorrance, W. H. and Dore, F. J.; "The Effect of Mass Transfer on the Compressible Turbulent Boundary Layer Skin Friction and Heat Tra 'fer," Journal of the Aerospace Sciences, Vol. 21, 1954, p. 404.
- 94. Kutateladze, S. S. and Leontzv, A. I., <u>Turbulent Boundary Layers in</u> <u>Compressible Gases</u>, Academic Press, New York, 1964.